## SECURITIES AND EXCHANGE COMMISSION

## FORM SC 13D

Schedule filed to report acquisition of beneficial ownership of 5% or more of a class of equity securities

Filing Date: **2024-11-21 SEC Accession No.** 0001104659-24-121207

(HTML Version on secdatabase.com)

## SUBJECT COMPANY

## **ZEEKR Intelligent Technology Holding Ltd**

CIK:1954042| IRS No.: 000000000 | State of Incorp.:E9 | Fiscal Year End: 1231 Type: SC 13D | Act: 34 | File No.: 005-94761 | Film No.: 241482029

SIC: 3711 Motor vehicles & passenger car bodies

Mailing Address RM2301 BLDG1, DADAO WANGCHAO SHANGWU CTR YINGFENG ST, XIAOSHAN DISTRICT. HANGZHOU

YINGFENG ST, XIAOSHAN YII DISTRICT, HANGZHOU DISTRICT, HANGZHOU ZHEJIANG PROVINCE F4 00000 00

Business Address RM2301 BLDG1, DADAO WANGCHAO SHANGWU CTR YINGFENG ST, XIAOSHAN DISTRICT, HANGZHOU ZHEJIANG PROVINCE F4 00000 86 400 003 6036

## **FILED BY**

#### **GHGK Innovation Ltd**

CIK:2042890| IRS No.: 000000000 | State of Incorp.:D8 | Fiscal Year End: 1231 Type: SC 13D

Mailing Address
ZEEKR BUILDING (DADAO
WANGCHAO
SHANGWU CENTRE),
XIAOSHAN DISTRICT
HANGZHOU F4 311200

Business Address
ZEEKR BUILDING (DADAO
WANGCHAO
SHANGWU CENTRE),
XIAOSHAN DISTRICT
HANGZHOU F4 311200
400-003-6036

# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

#### SCHEDULE 13D

**Under the Securities Exchange Act of 1934** 

## **ZEEKR Intelligent Technology Holding Limited**

(Name of Issuer)

Ordinary Shares, par value \$0.0002 per share (Title of Class of Securities)

98923K103<sup>(1)</sup> (CUSIP Number)

Mr. Cheung Chung Yan, David Room 2301, 23rd Floor, Great Eagle Centre 23 Harbour Road, Wan Chai, Hong Kong, PRC. (852) 2598 3333

(Name, Address and Telephone Number of Person Authorized to Receive Notices and Communications)

November 14, 2024
(Date of Event Which Requires Filing of this Statement)

If the filing person has previously filed a statement on Schedule 13G to report the acquisition that is the subject of this Schedule 13D and is filing this schedule because of \$\$240.13d-1(e), 240.13d-1(f) or 240.13d-1(g), check the following box.  $\square$ 

**Note:** Schedules filed in paper format shall include a signed original and five copies of the schedule, including all exhibits. See §240.13d-7 for other parties to whom copies are to be sent.

\* The remainder of this cover page shall be filled out for a reporting person's initial filing on this form with respect to the subject class of securities, and for any subsequent amendment containing information which would alter disclosures provided in a prior cover page.

The information required on the remainder of this cover page shall not be deemed to be "filed" for the purpose of Section 18 of the Securities Exchange Act of 1934 ("Act") or otherwise subject to the liabilities of that section of the Act but shall be subject to all other provisions of the Act (however, see the Notes).

(1) There is no CUSIP number assigned to the Ordinary Shares of the Issuer. CUSIP number 98923K103 has been assigned to the American Depositary Shares ("ADSs") of the Issuer, which are quoted on the New York Stock Exchange under the symbol "ZK." Each ADS represents ten Ordinary Shares of the Issuer.

CUSIP: 98923K103

1	Names of I	Names of Reporting Persons				
1 2 3 4 5 6 Numb Shares Benefi Owner Each Repor Persor With	Geely Auto	Geely Automobile Holdings Limited				
,	Check the Appropriate Box if a Member of a Group					
2 3 4 5 6 Numb Share Benet Owne Each Repor	(a)  (b)	X				
3	SEC Use C	nly				
4	Source of Funds (See Instructions)					
4	WC					
5	Check Box	if Dis	closure of Legal Proceedings Is Required Pursuant to Items 2(d) or 2(e)			
3						
(	Citizenship	or Pla	ce of Organization			
0	Cayman Is	lands				
		7	Sole Voting Power			
<b>N</b> T 1	C	7	0			
Share	s	8	Shared Voting Power			
			1,668,996,860 Ordinary Shares <sup>(1)</sup>			
	rtina		Sole Dispositive Power			
Perso			0			
With		10	Shared Dispositive Power			
		10	1,668,996,860 Ordinary Shares <sup>(1)</sup>			
	Aggregate	Amou	nt Beneficially Owned by Each Reporting Person			
11	1,668,996,860 Ordinary Shares <sup>(1)</sup>					
10	Check if the Aggregate Amount in Row (11) Excludes Certain Shares (See Instructions)					
12						
10	Percent of	Class I	Represented by Amount in Row (11)			
13	66.6% (2)	66.6% <sup>(2)</sup>				
	Type of Re	porting	g Person (See Instructions)			
14	CO	CO				

Includes (i) 1,368,996,860 Ordinary Shares of the Issuer directly held by Luckview Group Limited, a limited company incorporated (1) in the British Virgin Islands wholly owned by this Reporting Person; and (ii) 300,000,000 Ordinary Shares being purchased by Luckview pursuant to the ZEEKR Sale and Purchase Agreement, details of which are set forth in Item 3.

The percentage is calculated based on a total of 2,507,346,254 Ordinary Shares of the Issuer issued and outstanding as reported in the final prospectus of the Issuer dated May 9, 2024 after taking into account the subsequent full exercise of over-allotment option).

(2) The ownership percentage of Geely Automobile Holdings Limited in certain releases and foreign regulatory filings was computed based on a total of 2,657,346,254 Ordinary Shares, including the 150,000,000 Ordinary Shares reserved under the Issuer's 2021 Share Incentive Plan.

#### CUSIP: 98923K103

1	Names of Reporting Persons					
1	Luckview Group Limited					
2	Check the Appropriate Box if a Member of a Group					
2	(a) □ (b) ⊠					
3	SEC Use C	Only				
1	Source of Funds (See Instructions)					
4	AF	AF				
4	Check Box if Disclosure of Legal Proceedings Is Required Pursuant to Items 2(d) or 2(e)					
5						
(	Citizenship	or Pla	nce of Organization			
6	British Vir	British Virgin Islands				
		7	Sole Voting Power			
NT 1	C	7	0			
Numb Share		8	Shared Voting Power			
Benef Owne	ficially ed by		1,668,996,860 Ordinary Shares <sup>(1)</sup>			
Each	·		Sole Dispositive Power			
Repor Perso		9	0			
With			Shared Dispositive Power			
		10	1,668,996,860 Ordinary Shares <sup>(1)</sup>			
	Aggregate	Amou	nt Beneficially Owned by Each Reporting Person			
11	1,668,996,	1,668,996,860 Ordinary Shares <sup>(1)</sup>				
	Check if the Aggregate Amount in Row (11) Excludes Certain Shares (See Instructions)					
12	□					
	Percent of Class Represented by Amount in Row (11)					
13	66.6% <sup>(2)</sup>					
	Type of Reporting Person (See Instructions)					
14	CO	CO				

- (1) Includes 300,000,000 Ordinary Shares being purchased by this Reporting Person pursuant to the ZEEKR Sale and Purchase Agreement, details of which are set forth in Item 3.
- The percentage is calculated based on a total of 2,507,346,254 Ordinary Shares of the Issuer issued and outstanding as reported in (2) the final prospectus of the Issuer dated May 9, 2024 (after taking into account the subsequent full exercise of over-allotment option). The ownership percentage of Geely Automobile Holdings Limited in certain releases and foreign regulatory filings was computed

based on a total of 2,657,346,254 Ordinary Shares, including the 150,000,000 Ordinary Shares reserved under the Issuer's 2021 Share Incentive Plan.

#### CUSIP: 98923K103

#### Item 1. Security and Issuer.

This Statement on Schedule 13D (this "Statement") relates to the ordinary shares, par value \$0.0002 per share (the "Ordinary Shares"), of ZEEKR Intelligent Technology Holding Limited, an exempted company incorporated under the laws of the Cayman Islands (the "Issuer"), whose principal executive offices are located at Room 2301, Building 1, Dadao Wangchao Shangwu Center, Yingfeng Street, Xiaoshan District, Hangzhou, Zhejiang Province, China.

The Issuer's American depositary shares (the "ADSs"), each representing ten Ordinary Shares, are listed on the New York Stock Exchange under the symbol "ZK." Unless the context requires otherwise, references to "Ordinary Shares" or "Shares" in this Statement include Ordinary Shares underlying the issued and outstanding ADSs.

#### Item 2. Identity and Background.

This Statement is being filed by the following persons:

- (1) Geely Automobile Holdings Limited ("Geely Auto"), a Cayman Islands incorporated company with limited liability and listed on The Stock Exchange of Hong Kong under stock code "0175"; and
- (2) Luckview Group Limited ("Luckview"), a limited company incorporated in the British Virgin Islands.

Each of the foregoing persons is referred to as a "Reporting Person" and collectively as the "Reporting Persons." Pursuant to a Joint Filing Agreement dated hereof between the Reporting Persons (the "Joint Filing Agreement"), the Reporting Persons have agreed to file this Statement jointly in accordance with the provisions of Rule 13d-1(k)(1) under the Securities Exchange Act of 1934, as amended (the "Act"). A copy of the Joint Filing Agreement is attached hereto as Exhibit 99.1.

Geely Auto is principally engaged in the research and development, manufacturing and trading of automobiles, automobile parts and related automobile components, and investment holding. The business address of Geely Auto is Room 2301, 23rd Floor, Great Eagle Centre, 23 Harbour Road, Wan Chai, Hong Kong, PRC. Information regarding each director and executive officer of Geely Auto is set forth on <u>Schedule I</u> attached hereto. The single largest shareholder of Geely Auto is Zhejiang Geely Holding Group Co., Ltd, a PRC limited liability company ("Geely Holding"). Geely Holding is controlled by Shufu Li ("Mr. Li"), a director, the founder, and chairman of the Issuer.

Luckview is primarily engaged in investment holding. The business address of Luckview is P.O. Box 3152, Road Town, Tortola, British Virgin Islands. Information regarding each director and executive officer of Luckview is set forth on <u>Schedule II</u> attached hereto. Luckview is wholly owned by Geely Auto.

Mr. Li, Geely Holding and some other associates of Mr. Li may be deemed to beneficially own the subject shares reported in Item 4 below. They are separately filing a Schedule 13D with the Securities and Exchange Commission in respect of their respective beneficial ownerships in the Shares.

During the last five years, none of the Reporting Persons and their respective directors and officers identified in <u>Schedule I</u> and <u>Schedule II</u> has been: (i) convicted in a criminal proceeding (excluding traffic violations or similar misdemeanors) or (ii) a party to a civil proceeding of a judicial or administrative body of competent jurisdiction and as a result of such proceeding was or is subject to a judgment, decree or final order enjoining future violations of, or prohibiting or mandating activities subject to, federal or state securities laws or finding any violation with respect to such laws.

#### Item 3. Source and Amount of Funds or Other Consideration.

On November 14, 2024, Luckview entered into a sale and purchase agreement (the "ZEEKR Sale and Purchase Agreement") with Geely International (Hong Kong) Limited ("GIHK"), a wholly owned subsidiary of Geely Holding, pursuant to which GIHK has agreed to sell, and Luckview has agreed to buy, 300,000,000 Ordinary Shares (the "Sale Shares") for an aggregate consideration of US\$806,100,000, or US\$26.87 per ADS (reflecting the 15-day volume weighted average price per ADS on the New York Stock Exchange for the 15-trading day period ending on (and including) the last trading day immediately prior to the date of the ZEEKR Sale and Purchase Agreement) (such transaction, the "Acquisition of ZEEKR Shares"), subject to the terms and conditions set forth therein. The consideration of the Acquisition of ZEEKR Shares will be funded by the internal cash reserve of Geely Auto and its subsidiaries.

#### CUSIP: 98923K103

The ZEEKR Sale and Purchase Agreement contains customary representations and warranties from each of GIHK (as seller) and Luckview (as buyer) and is subject to customary closing conditions for a transaction of this nature.

The description of the ZEEKR Sale and Purchase Agreement contained herein is qualified in its entirety by reference to Exhibit 99.2, which is incorporated herein by reference.

Luckview became a registered shareholder of the Issuer prior to the Issuer's initial public offering of ADSs in the United States (the "ZEEKR IPO") in May 2024. Immediately prior to the ZEEKR IPO, Luckview was the registered owner of 1,240,000,000 Ordinary Shares. In the ZEEKR IPO, Luckview subscribed for an additional 129,009,520 Ordinary Shares represented by 12,900,952 ADSs and disposed of 12,660 Ordinary Shares represented by 1,266 ADS on behalf of Geely Auto. The disposition was made to fulfill Geely Auto's assured entitlement obligation to minority shareholders under the Listing Rules of The Stock Exchange of Hong Kong.

#### **Item 4. Purpose of Transaction.**

The information set forth in Item 3 is hereby incorporated by reference in this Item 4.

This Statement is being filed in connection with the Acquisition of ZEEKR Shares. Taking into account the Reporting Persons' existing holding in the Issuer and the Sale Shares subject to the ZEEKR Sale and Purchase Agreement, following the completion of the Acquisition of ZEEKR Shares, the Reporting Persons will have the power to control and/or dispose Shares representing approximately 66.6% of the total voting power in general shareholders meeting of the Issuer.

The purpose of the Acquisition of ZEEKR Shares is to further demonstrate support for the ZEEKR brand, simplify the Issuer's shareholder structure and enhance influence over the Issuer's strategic direction, thereby facilitating allocation of strategic resources and implementation of future plans, as well as strengthen equity control over the Issuer and help mitigate possible equity dilution in the future.

Additionally, as disclosed on the Form 6-K filed by the Issuer on November 14, 2024, on such date the Issuer entered into strategic integration transactions (the "Strategic Integration Transactions") with certain other entities controlled by Mr. Li, including LYNK & CO Automotive Technology Co., Ltd. ("LYNK & CO"), a Chinese-foreign joint venture in China for which Geely Auto has a 50% indirect ownership interest. Upon the consummation of the Strategic Integration Transactions, LYNK & CO will be owned indirectly as to 51% by the Issuer and 49% by Geely Auto, respectively. The consummation of the Strategic Integration Transactions is subject to a number of contingencies that are beyond the control of the contract parties, including the approval by the independent shareholders of Geely Auto in accordance with the Listing Rules of The Stock Exchange of Hong Kong.

The descriptions of the Reporting Persons' involvement in the Strategic Integration Transactions are qualified in their entirety by reference to Exhibits 99.3 and 99.4, which are incorporated herein by reference.

Except as set forth in this Item 4 or Item 6 below, the Reporting Persons have no present plans or proposals that relate to or that would result in any of the actions specified in subparagraphs (a) through (j) of Item 4 of Schedule 13D of the Act. The Reporting Persons reserve the right to take such actions in the future as they deem appropriate, including changing the purpose described above or adopting plans or proposals with respect to one or more of the items described in subparagraphs (a) through (j) of Item 4 of Schedule 13D.

#### CUSIP: 98923K103

#### Item 5. Interest in Securities of the Issuer.

The information contained on each of the cover pages of this Statement and the information set forth or incorporated in Items 2, 3, 4, and 6 are hereby incorporated herein by reference.

- (a) (b). The responses of each Reporting Person to Rows (7) through (13) of the cover pages are hereby incorporated by reference in this Item 5. Except as otherwise stated herein, each Reporting Person expressly disclaims any beneficial ownership of the Ordinary Shares held by each other Reporting Person.
- (c). Except as disclosed in this Statement, none of the Reporting Persons has effected any transaction in the Ordinary Shares during the past 60 days.
- (d). Except as disclosed in this Statement, to the best knowledge of the Reporting Persons, no other person has the right to receive or the power to direct the receipt of dividends from, or the proceeds from the sale of, the Ordinary Shares beneficially owned by the Reporting Persons.
- (e). Not applicable.

Although Geely Auto is a listed company on The Hong Kong Stock Exchange and is managed separately from Mr. Li and his associates, the Reporting Persons, Mr. Li and his associates, and their respective affiliates may be deemed to be members of a "group" within the meaning of Section 13(d)(3) of the Act in respect of their respective interests in the Ordinary Shares (the "Geely Group"). The Geely Group may be deemed to beneficially own an aggregate of 1,950,996,860 Ordinary Shares, representing approximately 77.8% of the Shares issued and outstanding. Each of the Reporting Persons expressly disclaims beneficial ownership of the Ordinary Shares beneficially owned by other members of the Geely Group, to the extent such beneficial ownership would be attributable to the Reporting Persons.

The Shares reported on this Schedule 13D as being beneficially owned by the Reporting Persons have included the 300,000,000 Ordinary Shares being purchased in the ZEEKR Sale and Purchase Agreement. The Acquisition of the ZEEKR Shares has not been completed, is subject to customary closing conditions, and is expected to be completed in the first quarter of 2025.

The Reporting Persons note that releases and foreign regulatory filings of Geely Auto report their ownership percentage in the Issuer on a fully diluted basis, which results in percentages different from the amounts calculated in accordance with Rule 13d-3 under the Act. The ownership percentage of Geely Auto in those releases was computed based on a total of 2,657,346,254 Ordinary Shares, including the 150,000,000 Ordinary Shares reserved under the Issuer's 2021 Share Incentive Plan.

#### Item 6. Contracts, Arrangements, Understandings or Relationships with Respect to Securities of the Issuer.

The information set forth in Items 3 and 4 is hereby incorporated by reference in this Item 6.

Except as described above or elsewhere in this Statement or incorporated by reference in this Statement, there are no contracts, arrangements, understandings or relationships (legal or otherwise) between the Reporting Persons or, to the best of their knowledge, any other person with respect to any securities of the Issuer, including, but not limited to, transfer or voting of any securities, finder's fees, joint ventures, loan or option arrangements, puts or calls, guarantees of profits, division of profits or losses, or the giving or withholding of proxies.

#### Item 7. Material to be Filed as Exhibits.

Exhibit No.	Description
99.1	Joint Filing Agreement
99.2	ZEEKR Sale and Purchase Agreement
99.3	LYNK & CO Equity Transfer Agreement

#### **SIGNATURE**

After reasonable inquiry and to the best of its knowledge and belief, each of the undersigned certifies that the information set forth in this statement is true, complete and correct.

Dated: November 21, 2024

#### **Geely Automobile Holdings Limited**

By: /s/ Gui Sheng Yue

Name: Gui Sheng Yue

Title: CEO and executive director

#### **Luckview Group Limited**

By: /s/ Gui Sheng Yue

Name: Gui Sheng Yue Title: Director

#### **SCHEDULE I**

#### **Directors and Executive Officers of Geely Auto**

The names of the directors and the names and titles of the executive officers of Geely Auto and their principal occupations are set forth below. The business address of the directors and executive officers is Room 2301, 23rd Floor, Great Eagle Centre, 23 Harbour Road, Wan Chai, Hong Kong.

Name	Position	Present Principal Occupation	Citizenship	Shares Beneficially owned in the Issuer
Directors:				
Li Shu Fu	Executive Director and Chairman of the Board	Management	Chinese	1,950,996,860
Li Dong Hui, Daniel	Executive Director and Vice Chairman of the Board	Management	Chinese	20,000,000
Gui Sheng Yue	Executive Director and Chief Executive Officer	Management	Chinese	10,000,000
Gan Jia Yue	Executive Director	Management	Chinese	4,350,000
Wei Mei Executive Director		Management	Chinese	5,800,000
Mao Jian Ming, Moosa	Executive Director	Management	Chinese	0
An Qing Heng	Independent Non-executive Director	Director of the Advisory Committee of China Automotive Industry	Chinese	0

Wang Yang	Independent Director	Non-executive	Partner of Primavera Capital Group	Chinese	0
Gao Jie	Independent Director	Non-executive	CFO of Lightspeed China Partners	Chinese	0
Yu Li Ping, Jennifer	Independent Director	Non-executive	Chairwoman and founding partner of Auster Capital Partners Limited	Chinese	0
Zhu Han Song	Independent Director	Non-executive	Chairman of Seekers Capital (Jiangyin) Co., Ltd.	Chinese	0
Executive Officers:					
Gui Sheng Yue	Executive Dire		Management	Chinese	10,000,000

## **SCHEDULE II**

## **Directors and Executive Officers of Luckview**

The names of the directors and the names and titles of the executive officers of Luckview and their principal occupations are set forth below. The business address of the directors and executive officers is OMC Chambers, Wickhams Cay 1, Road Town, Tortola, British Virgin Islands.

Name	Position	Present Principal Occupation	Citizenship	Shares Beneficially owned in the Issuer	
Directors:					
Li Shu Fu	Director	Director	Chinese	1,950,996,860	
Gui Sheng Yue	Director	Director	Chinese	10,000,000	

#### JOINT FILING AGREEMENT

In accordance with Rule 13d-1(k)(1) under the Securities Exchange Act of 1934, as amended, the persons named below agree to the joint filing on behalf of each of them of a statement on Schedule 13D (including amendments thereto) with respect to the Ordinary Shares (including any such securities underlying American Depositary Shares) of ZEEKR Intelligent Technology Holding Limited and further agree that this Joint Filing Agreement be included as an Exhibit to such joint filings.

In evidence thereof, the undersigned, being duly authorized, have executed this Joint Filing Agreement as of November 21, 2024.

#### **Geely Automobile Holdings Limited**

By: /s/ Gui Sheng Yue

Name: Gui Sheng Yue

Title: CEO and executive director

#### **Luckview Group Limited**

By: /s/ Gui Sheng Yue

Name: Gui Sheng Yue Title: Director

#### Dated the 14th day of November 2024

#### GEELY INTERNATIONAL (HONG KONG) LIMITED

(as Seller)

and

#### **LUCKVIEW GROUP LIMITED**

(as Purchaser)

#### SALE AND PURCHASE AGREEMENT

relating to approximately 11.3% (on a fully-diluted basis) of the entire issued share capital in

#### ZEEKR INTELLIGENT TECHNOLOGY HOLDING LIMITED

THE SECURITIES OFFERED HEREBY ARE BEING OFFERED PURSUANT TO REGULATION S UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT")

THE SECURITIES OFFERED HEREBY MAY NOT BE OFFERED OR SOLD IN THE UNITED STATES OR TO U.S. PERSONS AS DEFINED IN REGULATION S UNLESS SUCH SECURITIES HAVE BEEN REGISTERED UNDER THE SECURITIES ACT OR AN EXEMPTION FROM THE REGISTRATION REQUIREMENTS OF THE SECURITIES ACT IS AVAILABLE.

THIS AGREEMENT is made on the 14th day of November 2024.

#### **BETWEEN**

- (1) Geely International (Hong Kong) Limited (吉利国际(香港)有限公司), a limited liability company incorporated in Hong Kong, whose registered office is at Unit 2204, 22/F, Lippo Centre, Tower 2, 89 Queensway, Hong Kong (the "Seller"); and
- (2) **Luckview Group Limited**, a limited liability company incorporated in the British Virgin Islands, whose registered office is at OMC Chambers. Wickhams Cay 1, Road Town, Tortola, British Virgin Islands (the "**Purchaser**").

(Each of the Seller and the Purchaser shall be referred to individually as a "Party" and collectively as the "Parties").

#### **WHEREAS**

- (A) ZEEKR Intelligent Technology Holding Limited (the "Company") is an exempted company incorporated in the Cayman Islands, whose shares are listed on The New York Stock Exchange ("NYSE") and trading under the symbol "ZK".
- (B) As at the date of this Agreement, the Seller is the legal and beneficial owner of 300,000,000 ordinary shares of the Company, representing approximately 11.3% (on a fully-diluted basis) of the total issued share capital of the Company.

- The Purchaser is a wholly-owned subsidiary of Geely Automobile Holdings Limited ("Geely Automobile Holdings"), a company incorporated in the Cayman Islands with limited liability whose shares are listed on the main board of HKEx (stock codes: 175 (HKD counter) and 80175 (RMB counter)).
- The Seller has agreed to sell to the Purchaser, and the Purchaser has agreed to purchase from the Seller the Sale Shares (as defined herein below) on the terms and conditions hereinafter mentioned and in compliance with and in reliance upon provisions of Regulation S ("Regulation S") under the Securities Act of 1933, as amended (the "Securities Act").

-1-

#### **NOW IT IS HEREBY AGREED** as follows:

#### 1. Definitions and Interpretation

1.1 In this Agreement (including the recitals), the following words and expressions shall have the following meanings unless the context requires otherwise:

"ADS(s)"

the American Depositary Share(s) issued pursuant to a deposit agreement between the Company and a depositary, each representing 10 shares of the Company, which are listed on the NYSE;

"Agreed Exchange Rate"

the Chinese RMB/USD spot foreign exchange rate, expressed as the number of Chinese RMB per one USD, which appears on the Bloomberg Screen BFIX Page for the "USD/CNH" currency pair under the caption of "MID" at 3:00 p.m. London time on the Completion Date; or in case of no exchange rate on Bloomberg on the Completion Date, then the rate shall be the one last published on Bloomberg before the Completion Date;

"Business Day"

a day on which licensed banks in Hong Kong, the British Virgin Islands and the US (as applicable) are open for business, and excluding Sundays and Saturdays;

"Completion"

completion of the sale and purchase of the Sale Shares in accordance with the terms of this Agreement;

"Completion Date"

a date falling no later than the fifth Business Day after all Conditions are satisfied or waived, or such other date as the Purchaser and the Seller shall agree in writing, on which Completion is to take place;

"Conditions"

the conditions precedent referred to in Clause 4.1;

"Consideration"

the consideration payable for the purchase of the Sale Shares set out in Clause 3.1;

"Encumbrances"

any mortgage, charge, pledge, lien, usufruct, hypothecation, option, right of first refusal, right of first offer, pre-emption right, equities, deed of trust, warrant, adverse claims, or other encumbrance, priority or security interest, over or in any property, assets or rights of whatsoever nature or interest or any agreement for any of the same (otherwise than arising by statute or operation of law) and "Encumber" and "Encumbered" shall be construed accordingly;

"HKEx"	The Stock Exchange of Hong Kong Limited;
"Hong Kong"	Hong Kong Special Administrative Region of the People's Republic of China;
"Listing Rules"	the Rules Governing the Listing of Securities on The Stock Exchange of Hong Kong Limited;
"Long Stop Date"	12 months from the date of this Agreement, or such later date as may otherwise be agreed between the Seller and the Purchaser in writing;
"person"	includes any individual, company, corporation, firm, partnership, joint venture, association, organisation or trust (in each case, whether or not having a separate legal personality);
"Purchaser's Warranties"	the agreements, obligations, warranties, representations and undertakings of the Purchaser contained Clauses 7.2 and 7.3 in this Agreement and "Purchaser's Warranty" shall be construed accordingly;
"RMB"	Renminbi, the lawful currency of the People's Republic of China;
"Sale Shares"	300,000,000 ordinary shares of the Company legally and beneficially owned by the Seller, representing approximately 11.3% (on a fully-diluted basis) of the entire issued shares of the Company as at the date of Completion;
"Securities Act"	the United States Securities Act of 1993, as amended;
"Seller's Warranties"	the agreements, obligations, warranties, representations and undertakings of the Seller contained in Clauses 6.2 to 6.6 in this Agreement and "Seller's Warranty" shall be construed accordingly;

-3-

"Share Registrar"	Ogier Global (Cayman) Limited, acting in its capacity as the share registrar of the Company;
"US"	the United States of America;
"USD" or "US\$"	United States dollars, the lawful currency of the US; and
" <mark>0/0</mark> "	per cent.

- 1.2 In this Agreement unless the context otherwise requires:
  - (a) references to Clause(s), Recital(s) and Schedule(s) are references to clause(s) and recital(s) of and schedule(s) to this Agreement;
  - (b) Schedule(s) form(s) part of this Agreement and shall have the same force and effect as if expressly set out in the body of this Agreement and any reference to this Agreement shall include the Schedule(s);

- references to writing shall include typewriting, printing, lithography, photography, telecopier and telex messages and any mode of reproducing words in a legible and non-transitory form;
- words herein importing the singular shall include the plural and vice versa and words importing any gender shall include (d) all genders and words importing person shall include any individual, company, corporation, firm, partnership, joint venture, association or trust (in each case, whether or not having a separate legal personality);
- (e) any phrase introduced by the terms "including", "include", "in particular" or any similar expression shall be construed as illustrative and shall not limit the sense of the words preceding those terms; and
- (f) all capitalized terms used in this Agreement and not otherwise defined therein shall have the meanings ascribed to them under Regulation S.
- References in this Agreement to any ordinance, enactment, rule, law, directive or regulation include such ordinance, enactment, rule, law, directive or regulation as modified, consolidated, extended or re-enacted and include subsidiary legislation made thereunder.
- Any document referred to as being "in the agreed form" shall mean a document in a form agreed by the Parties prior to Completion and, where appropriate, initialled by or on their behalf for identification purposes.
- In this Agreement clause headings and the index are inserted for reference only and shall not affect construction or interpretation of this Agreement.
- The definitions and designations adopted in the recitals and the Schedule(s) and introductory statements preceding this Clause and the Schedule(s) shall apply throughout this Agreement and the Schedule(s).

-4-

When calculating the period of time before which, within which or following which any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded and if the last day of such period is not a Business Day, the period shall end on the next succeeding Business Day.

#### 2. Sale and Purchase

- Subject to and upon the terms and conditions of this Agreement, the Seller shall sell (as the legal and beneficial owner) and the Purchaser shall purchase, the Sale Shares free from all Encumbrances with all benefits and rights hereafter attaching thereto including all dividends or distributions which may be paid (except for those have been declared but not paid prior to the Completion Date), declared or made in respect thereof at any time on or after the Completion Date.
- In the event there is any stock split, right issue, placing or other matter that may cause the number of shares of the Company on the Completion Date to differ from that of the effective date of this Agreement, then the number of shares as defined in the Sale Shares shall be adjusted accordingly.

#### 3. Consideration

- The purchase price per ADS shall be the 15-day volume weighted average price per ADS on the NYSE for the 15-trading day period ending on (and including) the last trading day immediately prior to the date of this Agreement. Accordingly, the Consideration payable for the Sale Shares shall be US\$806,100,000.
- The Parties agree that the Consideration as stipulated in Clause 3.1 above is denominated in USD, but shall be converted into RMB at the Agreed Exchange Rate for the Completion on the Completion Date. The Consideration received by the Seller shall be paid in RMB.

3.3	follows:	sideration shall be settled in full by the Purchaser by way of a bank transfer to the Seller's designated bank account a
Accou	nt Name	
Bank N	Name	:
Bank A	Address	:
Swift (	Code	:
Bank (	Code	:
Accou	nt No.	:

-5-

#### 4. Conditions Precedent to Completion

- 4.1 Completion shall be subject to and conditional upon the fulfilment or waiver of the following conditions at or prior to Completion:
  - the obtaining by the Seller of all necessary consents, authorisations and/or approvals (or, as the case may be, the relevant (a) waiver) of any kind (whether governmental or otherwise) in connection with the entering into and performance of the terms of this Agreement and/or the change of shareholders of the Company upon Completion;
  - the obtaining by the Purchaser of all necessary consents, authorisations and/or approvals (or, as the case may be, the relevant (b) waiver) of any kind (whether governmental or otherwise) in connection with the entering into and performance of the terms of this Agreement and/or the change of shareholders of the Company upon Completion;
  - (c) the Seller's Warranties remaining true and accurate in all material respects as of the Completion Date by reference to the facts and circumstances subsisting as at the Completion Date;
  - (d) the Purchaser's Warranties remaining true and accurate in all material respects as of the Completion Date by reference to the facts and circumstances subsisting as at the Completion Date;
  - (e) Geely Automobile Holdings having complied with the Listing Rules in respect of this Agreement and all other requirements as may be imposed by the HKEx as a condition to this Agreement;
  - there being no applicable law which prohibits, restricts or imposes conditions or limitations on, or is reasonably expected to operate to prohibit, restrict or impose conditions or limitations on, the consummation of the transaction contemplated under this Agreement; and
  - (g) the Seller shall have delivered to the Purchaser, and the Purchaser shall have delivered to the Seller a copy of the closing certificate certifying that each of the conditions has been satisfied or waived.
- 4.2 The Seller may in its absolute discretion waive the Conditions referred to in Clauses 4.1(b) and (d) either in whole or in part at any time on or before Completion by notice in writing to the Purchaser.
- 4.3 The Purchaser may in its absolute discretion waive the Conditions referred to in Clauses 4.1(a) and 4.1(c) either in whole or in part at any time on or before Completion by notice in writing to the Seller.
- The Parties shall cooperate fully and use their best endeavours to procure the satisfaction of the Conditions on or before Completion and keep each other informed of the progress of satisfying the Conditions. If any of the Conditions are not fulfilled (or waived by the Seller or the Purchaser in whole or in part (as the case may be)) in accordance with this Agreement on or before Completion, the Parties shall not be obliged to proceed to Completion and the provisions of this Agreement shall from such date cease to have any effect.

#### 5. <u>Completion</u>

- Subject to fulfilment or waiver of all the Conditions, Completion shall take place on the Completion Date at such place as the Parties may agree when all (and not part only) the acts and requirements set out in Clauses 5.2 and 5.3 shall be complied with.
- 5.2 At Completion, the Seller shall:
  - (a) deliver to the Purchaser duly completed and signed stock power forms, authorising and instructing the Share Registrar to transfer record ownership of the Sale Shares to the account of the Purchaser in the share register of the Company; and
  - procure the Company to deliver to the Purchaser a share certificate(s) in the name of the Purchaser in respect of the Sale Shares. Such share certificate shall bear a legend in substantially the form set forth below (in addition to any other legends required under other applicable Laws):

The shares evidenced by this certificate have not been registered under the Securities Act of 1933, as amended (the "Securities Act"), or the securities laws of any state of the United States or under the laws of any other jurisdiction but have been issued in reliance on an exemption from registration under United States securities laws contained in Regulation S under the Securities Act. The shares evidenced by this certificate may not be transferred, nor will any assignee or endorsee hereof be recognized as an owner hereof by the issuer for any purpose, unless either the transfer is made in accordance with Regulation S, unless a registration statement under the Securities Act with respect to such shares shall then be in effect or unless the availability of an exemption from registration with respect to any proposed transfer or disposition of such shares shall be established to the satisfaction of counsel for the issuer.

- 5.3 At Completion, the Purchaser shall pay to the Seller an amount equal to the Consideration in the manner set out in Clauses 3.2 and 3.3.
- None of the Parties shall be obliged to complete the sale and purchase of the Sale Shares unless the other Party complies fully with the requirements of Clauses 5.2 and 5.3 which are expressed to be the other Party's obligations.
- If Completion does not take place on the Completion Date (the "Intended Completion Date") due to the Purchaser or Seller failing to comply with any of its obligations under this Clause 5 (whether such failure by such Party amounts to a repudiatory breach or not), then the Seller may (in the case of a default by the Purchaser) or the Purchaser may (in the case of a default by the Seller) (the "Non-Defaulting Party") in their respective absolute discretion, by written notice to the Purchaser or the Seller (as the case may be) (the "Defaulting Party"), and without prejudice to any other rights:

-7-

- proceed to Completion on that date, to the extent that the Non-Defaulting Party is ready, able and willing to do so, and specify a later date by which the Defaulting Party shall be obliged to complete its relevant outstanding obligations;
- (b) elect to defer Completion to a Business Day no later than the Long Stop Date; or
- (c) terminate this Agreement.

#### 6. Seller's Warranties

The Seller represents, warrants and undertakes to and in favour of the Purchaser that each of the Seller's Warranties is true and accurate in all material respects at the date of this Agreement and will continue to be so on each day up to and including Completion with reference to the facts and circumstances from time to time applying and acknowledges that the Purchaser is relying upon such Seller's Warranties in entering into this Agreement.

- The Seller warrants that the Seller has the requisite power and authority to execute and deliver this Agreement and to perform its obligations under this Agreement, and the terms of this Agreement constitute legal, valid and binding obligations enforceable against it.
- 6.3 The Seller is the sole legal and beneficial owner of the Sale Shares, and is entitled to transfer the full legal and beneficial ownership in the Sale Shares to the Purchaser free from all Encumbrances, subject to the terms set out in this Agreement.
- 6.4 The execution and delivery of, and the performance of the Seller's obligations under this Agreement will not:
  - result in a breach of, or give rise to a default under any contract or other instrument to which it is a party or by which it is bound;
  - result in a breach of any applicable laws or regulations or any order, judgment or decree of any court, governmental (ii) agency or regulatory authority applicable to it or any of its assets or to which it is a party or by which it is otherwise bound; and
  - (iii) require it to obtain any third-party consent to sell the Sale Shares.
- The Seller has not created or attempted or agreed to create or permit to arise or exist any Encumbrance in any material respect over all or any part of the Sale Shares or any interest therein or otherwise assign, deal with or dispose of all or any part of the Sale Shares (except under or pursuant to this Agreement).

-8-

- 6.6 The Seller has not granted or agreed to grant any options or other right in respect of the Sale Shares to any person.
- The Purchaser shall be entitled to take action both before and after Completion in respect of any breach or non-fulfilment of any of the Seller's Warranties and Completion shall not in any way constitute a waiver of any right of the Purchaser.
- 6.8 Upon Completion, the Purchaser shall be entitled to all benefits and rights attaching to the Sale Shares, notwithstanding nonfulfilment of any of the Seller's Warranties.

#### 7. <u>Purchaser's Warranties</u>

- The Purchaser represents, warrants and undertakes to and in favour of the Seller that each of the Purchaser's Warranties is true and accurate in all material respects at the date of this Agreement and will continue to be so on each day up to and including Completion with reference to the facts and circumstances from time to time applying and acknowledges that the Seller is relying upon such Purchaser's Warranties in entering into this Agreement.
- The Purchaser warrants that it has the requisite power and authority to execute and deliver this Agreement and to perform its obligations under this Agreement, and the terms of this Agreement constitute legal, valid and binding obligations enforceable against it.
- 7.3 The execution and delivery of, and the performance of the Purchaser's obligations under this Agreement will not:
  - (i) result in a breach of, or give rise to a default under any contract or other instrument to which it is a party or by which it is bound;
  - result in a breach of any applicable laws or regulations or any order, judgment or decree of any court, governmental agency or regulatory authority applicable to it or any of its assets or to which it is a party or by which it is otherwise bound; and
  - (iii) require it to obtain any third-party consent (except for those specified in Clause 4) to purchase the Sale Shares.

7.4 The Seller shall be entitled to take action both before and after Completion in respect of any breach or non-fulfilment of any of the Purchaser's Warranties and Completion shall not in any way constitute a waiver of any right of the Seller.

#### 8. <u>Confidentiality</u>

None of the Parties shall, without the prior written consent of the other Party, disclose the terms of, or any matters referred to in or any information relating thereto, this Agreement except to its professional advisers whose province it is to know such terms or matters or information and to those persons to whom it may be necessary to disclose such terms or matters or information for the purpose of or in connection with this Agreement and subject as required by law or by the competent regulatory authorities by virtue of any regulatory requirements.

-9-

None of the Parties shall make any public announcement in relation to the transactions the terms of which are set out in this Agreement or the transactions or arrangements hereby contemplated or herein referred to or any matter ancillary hereto or thereto without the respective prior written consents of the other Party (which consents shall not be unreasonably withheld or delayed) save as required by the competent regulatory authorities by virtue of any regulatory requirements. Except as may be required by law or by the competent regulatory authorities by virtue of any regulatory requirements, each Party shall also notify the other Party of any communications with the relevant competent regulatory authorities in connection with this Agreement.

#### 9. <u>Miscellaneous</u>

- 9.1 Fiduciary duties: Nothing in this Agreement shall be deemed to require the violation of the fiduciary duties of any director of any Party under applicable laws in the director's capacity as such.
- 9.2 No assignment: Each Party shall not assign or otherwise transfer any of its rights or obligations under this Agreement to any third party without the prior written consent of the other Party.
- 9.3 <u>Binding on successors</u>: This Agreement shall be binding upon and enforceable against the respective successors and assigns and representatives (as the case may be) of the Parties.
- Entire agreement: This Agreement (together with all agreements and documents executed contemporaneously with it or referred to in it) constitutes the entire agreement between the Parties in relation to the subject matter of it and supersedes all prior and any other commitments agreements promises or understandings whether oral or written with respect to that subject matter.
- 9.5 <u>Amendment</u>: This Agreement may only be amended or modified by the Parties in writing duly executed by the Parties.
- Partial invalidity: If any term, condition or provision of this Agreement is held to be a violation of any applicable law, statute or regulation it shall be deemed to be deleted from this Agreement and shall be of no force and effect and this Agreement shall remain in full force and effect as if that term, condition or provision had not originally been contained in this Agreement. Notwithstanding the foregoing, in the event of any such deletion the Parties shall negotiate in good faith in order to agree the terms of a mutually acceptable and satisfactory alternative provision in place of the provision so deleted.
- No waiver: No failure or delay on the part of any Party to exercise any right, power or remedy hereunder shall operate as a waiver thereof nor shall any single or partial exercise by any Party of any right, power or remedy hereunder preclude any other or future exercise thereof or the exercise of any other right, power or remedy. A waiver of any breach of this Agreement or any right of remedy under this Agreement shall not be effective, or implied, unless that waiver is in writing and is signed by the Party against whom that waiver is claimed.

9.8	<u>Rights cumulative</u> : The rights and remedies provided in this Agreement are cumulative and not exclusive of any rights or remedies provided by law.
10.	Costs and Expenses
	Save as otherwise provided in this Agreement, all expenses incurred by or on behalf of the Parties in connection with the negotiation, preparation or execution of this Agreement, shall be borne and paid solely by the Party who incurred the liability.
11.	<u>Notices</u>
11.1	Any notice or other communications to be given under this Agreement shall be in writing and shall be delivered by hand, sent by post or email. The details of the Parties in receiving notice or communications are as follows:
	To the Seller
Addres Email Attention	:
	To the Purchaser
Addres Email Attention	:
11.2	Any notice or communication, if delivered by hand, shall be deemed received when delivered at the relevant address; if sent by post, shall be deemed received 3 Business Days after the date of dispatch; and if sent by email, shall be deemed received at the time as recorded on the device from which the sender sent the email unless the sender receives an automated message that the email has not been delivered.
12.	Counterparts
	This Agreement may be executed in one or more counterparts, each of which so executed shall constitute an original and all of which together individually or otherwise executed by the Parties will constitute one and the same document.
13.	Third Party Rights
	A person who is not a Party has no right under the Contracts (Rights of Third Parties) Ordinance (Chapter 623 of the Laws of Hong Kong) to enforce or to enjoy the benefit of any term of this Agreement.
14.	Governing Law and Jurisdiction
	This Agreement is governed by and shall be construed in accordance with the laws of Hong Kong and the Parties hereby submit to the non-exclusive jurisdiction of the courts of Hong Kong.
	(The remainder of this page is intentionally left blank)
	-11-
AS WI	TNESS this Agreement has been duly executed by each of the Parties as of the date first above written.
The Se	
/s/ Li D	Oonghui )

for and on behalf of	)
GEELY INTERNATIONAL	)
(HONG KONG) LIMITED	)
in the presence of:	)
/s/ Li Wenqi	)
1760 Jiangling Road,	)
Hangzhou, China	)
The Purchaser	
SIGNED by	)
	)
/s/ Gui Sheng Yue	)
for and on behalf of	)
LUCKVIEW GROUP LIMITED	)
in the presence of:	)
/s/ Lin Jianyong	)
1760 Jiangling Road,	)
Hangzhou, China	)

#### **EQUITY TRANSFER AGREEMENT**

Regarding

#### LYNK & CO AUTOMOTIVE TECHNOLOGY CO., LTD.

by and among

#### ZHEJIANG ZEEKR INTELLIGENT TECHNOLOGY CO., LTD.

(as the Buyer)

and

#### ZHEJIANG GEELY HOLDING GROUP CO., LTD.

#### VOLVO CARS (CHINA) INVESTMENT CO., LTD.

(as the Sellers)

and

#### LYNK & CO AUTOMOTIVE TECHNOLOGY CO., LTD.

Dated November 14, 2024

This EQUITY TRANSFER AGREEMENT (this "Agreement") is entered into by and among the following parties on November 14, 2024 (the "Effective Date"):

- Zhejiang ZEEKR Intelligent Technology Co., Ltd. (浙江极氪智能科技有限公司), a limited liability company established (1) in the PRC with its address at Room 1031, Shangwu Building 1, No. 1388 Minshan Road, Beilun District, Ningbo, Zhejiang Province, the PRC (the "Buyer");
- Zhejiang Geely Holding Group Co., Ltd. (浙江吉利控股集团有限公司), a limited liability company established in the PRC with its address at 1760 Jiangling Road, Binjiang District, Hangzhou, Zhejiang Province, the PRC ("GH");
- Volvo Cars (China) Investment Co., Ltd. (沃尔沃汽车(中国)投资有限公司), a limited liability company established in the PRC with its address at Floor 1, Building 7, 2088 Lvyi Road, Jiading Industrial District, Shanghai, the PRC ("VCI", together with GH, the "Sellers", and each a "Seller"); and
- LYNK & CO Automotive Technology Co., Ltd. (领克汽车科技有限公司), a limited liability company established in the PRC with its address at 201-206, No. 918 Binhai Fourth Road, Andong Town, Qianwan New District, Ningbo, Zhejiang Province, the PRC (the "Company").

Each of the parties listed above is referred to hereinafter individually as a "Party" and collectively as the "Parties".

#### RECITALS

WHEREAS, as of the Effective Date, GH owns 20% of the issued and paid up equity interests in the Company (the "GH Equity"), and VCI owns 30% of the issued and paid up equity interests in the Company (the "VCI Equity", together with GH Equity, the "Target Equity");

**WHEREAS**, the Sellers desire to sell to the Buyer, and the Buyer desires to purchase from the Sellers, the Target Equity for the purchase price and upon the terms and conditions hereinafter set forth;

WHEREAS, concurrently with the execution of this Agreement, the Buyer, the Company and Ningbo Geely Automobile Industry Co., Ltd. (宁波吉利汽车实业有限公司) ("Ningbo Geely"), an owner of 50% of the issued and paid up equity interests in the Company, entered into a Subscription Agreement (the "Capital Injection Agreement"), pursuant to which the Buyer will subscribe for certain increased registered capital of the Company, upon terms and subject to conditions set forth therein (such transaction, the "Lynk Capital Injection");

**NOW, THEREFORE**, in consideration of the representations, warranties, covenants and agreements hereinafter contained, the Parties agree as follows:

#### 1. DEFINITIONS AND INTERPRETATION

1.1 Capitalized terms used in this Agreement shall have the following meanings.

1

- "Accounting Standards" means the generally accepted accounting principles in the PRC, the generally accepted accounting principles in the United States ("US GAAP") or the Hong Kong Financial Reporting Standards, as applicable.
- "Action" means any litigation, hearing, claim, suit, charge, action, complaint, arbitration, settlement, decision, inquiry, investigation or other proceedings initiated by or before any Governmental Authority.
- "Affiliate" of a Person means any other Person that, directly or indirectly, Controls, is Controlled by, or is under common Control with such Person; provided that, for purposes of this Agreement, (i) none of the Sellers or any Group Company shall be deemed an Affiliate of the Buyer, (ii) none of the ZEEKR Group, GH Group, Volvo Car Group and the Group Companies shall be deemed an Affiliate of each other, (iii) GH's Affiliate is any member of GH Group, and VCI's Affiliate is any member of Volvo Car Group, the Affiliate of the Buyer is any member of ZEEKR Group.
  - "Agreement" has the meaning ascribed thereto in the Preamble.
  - "Amended Articles" has the meaning ascribed thereto in Section 3.2.3(a).
  - "AMR" means the State Administration for Market Regulation or its competent local counterparts.
  - "AMR Registration" has the meaning ascribed thereto in Section 4.2.6.
- "Anti-Money Laundering Law" means applicable anti-money laundering statutes of all jurisdictions where the Group Companies conduct business, the rules and regulations thereunder and any related or similar rules, regulations or guidelines, issued, administered or enforced by any Governmental Authority.
- "Anticorruption Law" means Laws relating to anti-bribery or anticorruption (governmental or commercial), which apply to the business and dealings of any Group Company, including Laws that prohibit the corrupt payment, offer, promise or authorization of the payment or transfer of anything of value (including gifts or entertainment), directly or indirectly, to any Government Official, government employee or commercial entity to obtain or retain business or a business advantage.
  - "Basket Amount" has the meaning ascribed thereto in Section 8.2.4(a).

"Business Day" means any day that is not a Saturday, a Sunday or other day on which commercial banks are required or authorized by Law to be closed in the PRC.

"Buver" has the meaning ascribed thereto in the Preamble.

"Buyer Cap" has the meaning ascribed thereto in Section 8.3.4.

"Capital Injection Agreement" has the meaning ascribed thereto in the Recitals.

"CIETAC" has the meaning ascribed thereto in Section 11.2.2.

2

"Closing" has the meaning ascribed thereto in Section 3.1.

"Closing Date" has the meaning ascribed thereto in Section 3.1.

"Company" has the meaning ascribed thereto in the Preamble.

"Company Cap" has the meaning ascribed thereto in Section 8.2.4(c).

"Company Securities" has the meaning ascribed thereto in Section 2 of Schedule II.

"Company Subsidiary Securities" has the meaning ascribed thereto in Section 3 of Schedule II.

"Confidential Information" has the meaning ascribed thereto in Section 9.1.

"Contract" means any contract, agreement, indenture, note, bond, loan, instrument, lease, mortgage, franchise, license, commitment, purchase order, and other legally binding arrangement, whether written or oral.

"Control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management of a Person, whether through the ownership of voting securities, by Contract, credit arrangement or proxy, as trustee, executor or agent or otherwise. For purposes of this definition, a Person shall be deemed to Control another Person if such first Person, directly or indirectly, owns or holds more than 50% of the voting equity securities in such other Person for the general election of directors, or if such first Person, directly or indirectly, controls the board of directors, managing partner or other similar governing body or position of such other Person. The terms "Controlled" and "Controls" shall have meanings correlative to the foregoing.

"Disclosure Letter" has the meaning ascribed thereto in Schedule II.

"**Dispute**" has the meaning ascribed thereto in Section 11.2.1.

"Effective Date" has the meaning ascribed to it in the Preamble.

"Encumbrance" means any lien, pledge, hypothecation, mortgage, security interest, charge, claim, title retention, levy, proxy, option, restrictive covenant, license, right of first refusal, right of first offer, easement, or other encumbrance or adverse claim of any kind (including without limitation any conditional sale agreement, capital lease or other title retention agreement relating to property or asset).

"Equity Transfer" has the meaning ascribed thereto in Section 2.1.

"Environmental Laws" means any Laws pertaining to: (a) the protection of the environment (including air quality, surface water, groundwater, soils, subsurface strata, sediments, drinking water, noise, natural resources and biota); (b) the protection of human health and safety or natural resources, but only with respect to exposure to Hazardous Materials; or (c) the use, registration, management, generation, storage, treatment, recycling, disposal, discharge, transport, release, threatened release, investigation or remediation of Hazardous Materials.

"Financial Statements" has the meaning ascribed to it in Section 5 of Schedule II.

"<u>Fundamental Representations</u>" means the representations and warranties set forth in Schedule I and Sections 1, 2, 3, 4 and 6 of Schedule II.

"<u>GA</u>" means Geely Automobile Holdings Limited (吉利汽車控股有限公司), a company established in the Cayman Islands with its shares listed on the Main Board of the Stock Exchange of Hong Kong Limited.

"GH" has the meaning ascribed thereto in the Preamble.

"GH Equity" has the meaning ascribed thereto in the Recitals.

"GH Group" means, collectively, GH and its Subsidiaries currently and hereafter, but, for the purpose of this Agreement, excluding ZEEKR Group and Volvo Car Group.

"Government Official" means (a) any official, officer, employee or representative of, or other individual acting for or on behalf of, any Governmental Authority or agency or instrumentality thereof (including any state-owned or controlled enterprise), or any public international organization (as defined in the U.S. Foreign Corrupt Practices Act), (b) any political party or party official or candidate for political office or (c) any company, business, enterprise or other entity owned, in whole or in part, or controlled by any Person described in the foregoing clause (a) or (b) of this definition.

"Governmental Authority" means any nation or government or any province or state or local or any other political subdivision thereof, or any entity, authority or body exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government, including any government authority, agency, department, board, commission or instrumentality or any political subdivision thereof, any court or arbitrator, any arbitration tribunal, and any self-regulatory organization or national or international stock exchange.

"Governmental Order" means any applicable order, ruling, decision, verdict, decree, writ, subpoena, mandate, precept, command, directive, consent, approval, award, judgment, injunction or other similar determination or finding by, before or under the supervision of any Governmental Authority.

"Grants" has the meaning ascribed to it in Section 17 of Schedule II.

"Group Companies" means collectively, the Company and its Subsidiaries currently and hereafter, and each is herein referred to individually as a "Group Company".

"<u>Hazardous Materials</u>" means any substance that has been designated by any Governmental Authority or by applicable Laws to be radioactive, toxic, hazardous or otherwise a danger to health or the environment.

"Intellectual Property" means any and all (a) patents (including all reissues, divisionals, provisionals, continuations, continuations in part, re-examinations, renewals and extensions thereof), patent applications, and other patent rights, (b) trademarks, service marks, tradenames, brand names, logos, slogans, trade dress, design rights, and other similar designations of source or origin, together with all goodwill associated with any of the foregoing and applications, registrations and renewals in connection therewith, (c) copyrights, mask works, and copyrightable works, and all applications, registrations for and renewals in connection therewith, (d) internet domain names, web addresses, web pages, websites and related content, accounts with social media companies and the content found thereon and related thereto, and uniform resource locators, (e) proprietary Software, including source code, object code and supporting documentation for such Software, (f) trade secrets and proprietary information, including confidential business information, technical data, customer lists, data collections, methods and inventions (whether or not patentable and where or not reduced to practice), (g) copies and tangible embodiments of any of the foregoing, (h) all other intellectual property, whether or not registrable, in each case, under any Law or statutory provision or common law doctrine in any country, and (i) all rights to sue or recover and retain damages and costs and attorneys' fees for past, present and future infringement or misappropriation of any of the foregoing.

"Indemnified Party" or "Indemnified Parties" has the meaning ascribed thereto in Section 8.2.1.

"Indemnifying Party" has the meaning ascribed thereto in Section 8.2.1.

"<u>Information Technology</u>" means all computer systems, telecommunication systems, software (and the tangible media on which it is stored) and hardware including source and object code, cabling, routers, switched, racks, servers, PCs, laptops, terminals, scanners, printers and all associated peripherals, excluding in all cases Intellectual Property.

"Interim Period" has the meaning ascribed thereto in Section 6.1.

"Joint Venture Contract" has the meaning ascribed thereto in Section 3.4.

"Key Person" means officers, directors, and any persons with managing or operational control.

"Law(s)" means any federal, state, territorial, foreign or local law, common law, statute, ordinance, rule, regulation, code, constitution, treaty, measure, notice, circular, judgment, decree, opinion, Governmental Order or other requirement or rule of law of any Governmental Authority, including any rules promulgated by a stock exchange or regulatory body.

"Leakage" means any of the following, but in each case, excluding any Permitted Leakage:

- any dividend or other distribution (whether in cash or in specie) declared, paid or made, in each case by any Group (i) Company in favor of any Seller or (with respect to GH) any member of GH Group or (with respect to VCI) any member of Volvo Car Group (excluding the Group Companies in each case) (such Seller's "Seller Group");
- any return of capital or any payment of principal of, or interest on, any loan or other debt obligation, except for the (ii) repayment of borrowings that have been reflected in the Locked Box Account and have become due during the Locked Box Period pursuant to the terms of such borrowings;

5

- (iii) any Group Company paying, incurring or otherwise assuming liability for any fees, costs or expenses of the Sellers in connection with the transactions contemplated by this Agreement;
- (iv) any release or waiver in favor of any Seller Group by any Group Company of (i) any liability owed by such Seller Group or (ii) any claim against such Seller Group;
- (v) any payment of any nature (including bonuses or other compensation of any kind) by any Group Company for the benefit of any Seller Group, or any senior managements, officers or directors of any Sellers Group;
- (vi) any transfer or surrender of assets to, or liabilities assured, guaranteed, indemnified, secured or incurred by any Group Company for the benefit of any Seller Group;
- (vii) any agreement or arrangement to give effect to any of the matters referred to in paragraphs (i) to (vi) above; and
- (viii) any tax paid or payable by any Group Company as a result of any of the matters referred to in paragraphs (i) to (vi) above;

in each case net of: (i) any amount in respect of VAT which is recoverable as input tax by a Group Company; and (ii) with respect to such Group Company, any relief that arises or is available in connection with (i) to (viii) above.

"<u>Leased Real Property</u>" means the real properties leased, subleased, licensed or otherwise occupied by any Group Company as tenant, sublessee, licensee or occupier, together with, to the extent leased by any Group Company, all buildings and other structures, facilities, improvements or fixtures currently or hereafter located thereon.

"Leases" means all leases, subleases, licenses, concessions and other agreements (written or oral), including all amendments, extensions, renewals, guarantees and other agreements with respect thereto, pursuant to which any Group Company holds any Leased Real Property, including the right to all security deposits and other amounts and instruments deposited by or on behalf of any Group Company thereunder.

"Licensed Trademarks" has the meaning ascribed thereto in Section 6.6.2.

"<u>Listed Person</u>" means (i) any Person designated for trade sanctions or export control restrictions on a list published by the European Union, United States, United Nations or other relevant country or authority, or otherwise subject to such trade sanctions or export control restrictions, (ii) the Government of Venezuela, and (iii) companies, entities or organizations that are owned 50 percent or greater by any combination of Listed Persons or controlled by a Listed Person.

"<u>Liability(ies)</u>" means any and all debts, taxes, liabilities and obligations, whether accrued, fixed, absolute or contingent, whether due or to become due, whether determined or determinable, including the liabilities arising under any law, Action or Governmental Order or any contract, agreement, arrangement, commitment or covenant, and including any guarantee provided for the benefit of any bank or third party and the provisions accrued or recognized in accordance with the Accounting Standards.

6

"<u>Licensed Intellectual Property</u>" means all Intellectual Property owned by any Person other than a Group Company and licensed or sublicensed to any Group Company, or for which any Group Company has obtained a covenant not to be sued.

"Locked Box Accounts" means the unaudited consolidated financial statements of the Company and its Subsidiaries as of the Locked Box Date, a copy of which is included in Schedule IV hereto.

"Locked Box Claim" means a claim for payment pursuant to Section 2.3 of this Agreement.

"Locked Box Date" means September 30, 2024.

"Locked Box Interest Rate" means a rate per annum of 3.1% (on the basis of a 365-day year).

"Locked Box Period" means the period from but excluding the Locked Box Date and ending on and including the Closing Date.

"Long Stop Date" has the meaning ascribed thereto in Section 3.3.

"Loss(es)" means, without duplication, any and all demands, losses, damages, penalties, judgments, fines, taxes, claims, liabilities and obligations of any kind or nature whatsoever, including costs and expenses of investigating, proceedings, preparing or defending any such claim or Action and legal fees and expenses in connection therewith, and any amount paid in settlement of, any pending or threatened Action, but in each case excluding indirect, incidental, special, exemplary, punitive, consequential loss, any loss of profits, loss of production, loss of revenue, business interruption, loss of data or loss of business information arising out of, based on or resulting from this Agreement.

"LPR" means the RMB Loan Prime Rate announced by the National Interbank Funding Center (全国银行间同业拆借中心).

"Lynk Capital Injection" as the meaning ascribed thereto in the Recitals.

"Material Adverse Effect" means any event, fact, circumstance, occurrence, development, change or effect that, individually or in the aggregate with all other events, facts, circumstances, occurrences, developments, changes or effects, has, or would reasonably be expected to have a material adverse effect on (a) the business, conditions, assets, liabilities (including but not limited to contingent liabilities), operations or financial conditions of the Group Companies, taken as a whole; or (b) the qualifications or abilities of the Group Companies, taken as a whole, in carrying out the business currently engaged by the Group Companies; or (c) the ability of any of the Sellers to consummate the transactions contemplated by this Agreement; provided, however, that in determining whether a Material Adverse Effect has occurred pursuant to sub-clauses (a) or (b) above, there shall be excluded any effect on the Group Companies to the extent arising out of any of the following: (i) any action required to be taken pursuant to the terms and conditions of this Agreement or with the written consent of the Buyer; (ii) the announcement of the transactions contemplated by this Agreement; (iii) any changes

in applicable Laws or the interpretation or enforcement thereof or in applicable accounting principles or the interpretation thereof; or (iv) any pandemic, earthquake, typhoon, other natural disasters, any outbreak or escalation of hostilities of war or any act of terrorism; except in the case of sub-clauses (iii) or (iv), to the extent that the Group Companies, taken as a whole, are disproportionately affected thereby as compared with other participants in the same industries and geographic markets in which the Group Companies operate.

7

"Material Company Permits" has the meaning ascribed thereto in Section 14(b) of Schedule II.

"Material Contracts" has the meaning ascribed thereto in Section 12(b) of Schedule II.

"Ningbo Geely" has the meaning ascribed thereto in the Recitals.

"Owned Intellectual Property" means all Intellectual Property owned or purported to be owned by the Group Companies.

"Owned Real Property" means all land, together with all buildings, structures, improvements and fixtures located thereon, owned by any Group Company.

"Party" or "Parties" has the meaning ascribed thereto in the Preamble.

"Permitted Leakage" means any of the following:

- (a) any payments made (or to be made) by the Group Companies which have been specifically accrued or provided for in the Locked Box Accounts;
- any payments made (or to be made) by any of the Group Companies pursuant to existing agreements; provided that any (b) such payments are (A) made or arise in the ordinary course of business and consistent with past practice on an arm's length basis and (B) reasonably necessary for the operation of the business of the Group Companies;
- any payments in respect of salaries, directors' fees, pension contributions, expenses or bonuses made to, or in respect of services provided by, employees, workers, directors, officers or consultants of the Group Companies which are made (or to be made) by the Group Companies in the ordinary course of business and in accordance with the terms of the related employment or service contract;
- (d) any other payment, accrual, transfer of assets or assumption of liability by the Group Companies which the Buyer has expressly approved in writing; and
- (e) any tax payable by any Group Company as a consequence of any of the matters referred to in any of the sub-clauses (a) to (d).

"Permitted Encumbrances" means (i) statutory liens for current taxes, special assessments or other governmental or quasi-governmental charges not yet delinquent or the amount or validity of which is being contested in good faith by appropriate proceedings, (ii) mechanics', materialmen's, carriers', workers', warehousemen's, repairers' and similar statutory liens arising or incurred in the ordinary course of business, (iii) zoning, entitlement, building and other land use regulations imposed by any Governmental Authority, (iv) covenants, conditions, restrictions, easements and other similar matters of record affecting title to any real property which do not materially impair the occupancy or use of such real property for the purposes for which it is currently used or proposed to be used in connection with the Group Companies' businesses, (v) liens incurred or deposits or pledges made in connection with, or to secure payment of, worker's compensation, unemployment insurance, old age pension programs, social security, retirement and similar laws, and (vi) non-exclusive licenses of Intellectual Property entered into in the ordinary course of business.

"Person" means an individual, corporation, limited liability company, partnership, company, joint venture, association, trust, unincorporated organization, Governmental Authority, or other entity.

"PRC" means the People's Republic of China, for purpose of geographical and jurisdictional reference in this Agreement, excluding the Hong Kong Special Administrative Region, the Macau Special Administrative Region and Taiwan.

"Prohibited Person" means any Person that is (a) a national of any Sanctioned Country, (b) listed on the United States Commerce Department's Denied Parties List, Entity List, or Unverified List, the U.S. Department of Treasury's Office of Foreign Assets Control ("OFAC") Specially Designated Nationals and Blocked Persons List, Specially Designated Narcotics Traffickers or Specially Designated Terrorists, the Annex to Executive Order No. 13224, or any other sanctioned party list administered by OFAC, the Department of State's Debarred List, or UN Sanctions, or (c) any entity that is, in the aggregate, fifty percent (50%) or greater owned, directly or indirectly, or otherwise controlled by a Person or Persons described in clause (b).

"Purchase Price" has the meaning ascribed thereto in Section 2.2.1.

"Qualifying Claim" has the meaning ascribed thereto in Section 8.2.4(a).

"Real Property" means collectively the Owned Real Property and the Leased Real Property.

"Representatives" has the meaning ascribed thereto in Section 9.1.

"RMB" means Renminbi, the lawful currency of the PRC.

"Sanctioned Countries" means any country or territory subject to comprehensive, government-wide, or broad sectoral sanctions as may be in place or imposed from time to time, including - but not limited to - Belarus, Cuba, Iran, North Korea, Russia, Syria, Venezuela, or the Crimea, Donetsk, Luhansk, Kherson, or Zaporizhzhia regions of Ukraine.

"Sanctions" means the economic or financial sanctions laws, regulations, trade embargoes, export controls or other restrictive measures enacted, administered, implemented, and/or enforced from time to time by the European Union, the United Nations or the United States or other relevant country or authority.

"Seller" or "Sellers" has the meaning ascribed thereto in the Preamble.

"Seller Basket Amount" has the meaning ascribed thereto in Section 8.3.3.

9

"Seller Cap" has the meaning ascribed thereto in Section 8.2.4(b).

"Seller Indemnified Party" or "Seller Indemnified Parties" has the meaning ascribed thereto in Section 8.3.2.

"Seller Qualifying Claim" has the meaning ascribed thereto in Section 8.3.3.

"Short Form Agreement" has the meaning ascribed thereto in Section 13.7.

"Software" means any and all (a) computer programs, applications, systems and software, including any and all software implementations of algorithms, models and methodologies and any and all source code, object code, development and design tools, applets, compilers and assemblers, (b) databases and compilations, including any and all libraries and collections of data whether machine readable or otherwise, (c) descriptions, flow-charts and other work product used to design, plan, organize and develop any of the foregoing, (d) technology supporting, and the contents and audiovisual displays of, any internet site(s), and (e) documentation and media, including user manuals and training materials, relating to or embodying any of the foregoing or on which any of the foregoing recorded.

"<u>Subsidiary</u>" means, with respect to any Person, any corporation, partnership, limited liability company or other organization, whether incorporated or unincorporated, which is Controlled by such Person.

- "Target Equity" has the meaning ascribed thereto in the Recitals.
- "<u>VCI</u>" has the meaning ascribed thereto in the Preamble.
- "VCI Equity" has the meaning ascribed thereto in the Recitals.
- "Volvo Car Group" means Volvo Car Corporation AB and its Subsidiaries currently and hereafter.
- "ZEEKR" means ZEEKR Intelligent Technology Holding Limited, a company established in the Cayman Islands with its shares listed on the New York Stock Exchange.
  - "ZEEKR Group" means, collectively, ZEEKR and its Subsidiaries currently and hereafter.
- 1.2 Headings are inserted for convenience only and shall not affect the construction of this Agreement.
- 1.3 Unless the express context otherwise requires, references to:
  - Sections and Schedules are sections in and schedules to this Agreement and the Recitals and Schedules to this Agreement shall be deemed to form part of this Agreement;
  - 1.3.2 writing shall include any methods of producing or reproducing words in a legible and non-transitory form;

10

- 1.3.3 the singular number shall include the plural and vice versa;
- 1.3.4 "days" shall mean calendar days unless Business Days are expressly specified;
- 1.3.5 any Person shall include such Person's successors and permitted assigns; and
- any agreement or instrument shall be construed as a reference to that agreement or instrument as amended, novated or supplemented.
- The words "hereof," "hereby," "herein," and "hereunder" and words of similar import, when used in this Agreement, shall refer to this Agreement as a whole and not to any particular provision of this Agreement.
- Wherever the word "include," "includes" or "including" is used in this Agreement, it shall be deemed to be followed by the words "without limitation".
- 1.6 The word "or" shall not be exclusive.
- The word "extent" and the phrase "to the extent" shall mean the degree to which a subject or other thing extends, and such word or phrase shall not simply mean "if".
- In construing this Agreement, general words shall not be given a restrictive meaning by reason of the fact that they are followed by examples intended to be embraced by the general words.
- References to statutory provisions, including any subordinate legislation made under the relevant statute, shall be construed as references to those provisions as amended or re-enacted or as their application is modified by other provisions (whether before or after the date hereof) from time to time and shall include any provisions of which they are re-enactment (whether with or without modification).
- When calculating the period of time before which, within which or following which any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded and if the last day of such period is not a Business Day, the period shall end on the next succeeding Business Day.

The Parties have each participated in the negotiation and drafting of this Agreement and if an ambiguity or question of interpretation should arise, this Agreement shall be construed as if drafted jointly by the Parties and no presumption of burden of proof shall arise favoring or burdening any Party by virtue of the authorship of any provision in this Agreement.

#### 2. SALE AND PURCHASE

#### 2.1 Sale and Purchase

On the terms and subject to the conditions contained herein, each Seller agrees to sell, transfer and assign to the Buyer, and the Buyer agrees to acquire and purchase from each Seller, all of the rights, titles, and interests in and to the Target Equity owned by each such Seller, free and clear of any Encumbrance at the Closing (the "Equity Transfer").

11

For the avoidance of doubt, from and after the Closing Date, the Buyer shall be entitled to all of the rights, titles, and interests in and to the Target Equity.

#### 2.2 Purchase Price

- The aggregate purchase price to be paid by the Buyer as consideration for the Target Equity shall be RMB 9,000,000,000 *plus* the interest accrued on such amount during the Locked Box Period calculated at the Locked Box Interest Rate, among which,
  - (i) the purchase price to be paid by the Buyer to GH for the transfer of GH Equity shall be RMB3,600,000,000 *plus* the interest accrued on such amount during the Locked Box Period calculated at the Locked Box Interest Rate; and
  - (ii) the purchase price to be paid by the Buyer to VCI for the transfer of VCI Equity shall be RMB5,400,000,000 *plus* the interest accrued on such amount during the Locked Box Period calculated at the Locked Box Interest Rate.

(each, such Seller's "Purchase Price").

Each Seller will be responsible for clearing all its own tax liabilities and filing obligations in connection with its Equity 2.2.2 Transfer and the Buyer should hold no responsibility for such taxes and any other related costs payable by the Sellers under the applicable Laws.

#### 2.3 Locked Box

Each Seller undertakes to pay to the Buyer on demand an amount in cash equal to the amount of any Leakage received by such Seller or any member of such Seller's Seller Group during the Locked Box Period. The maximum liability of each Seller in respect of a Locked Box Claim shall not exceed the amount of Leakage received by such Seller and members of such Seller's Seller Group giving rise to such Locked Box Claim, provided that nothing in this Section 2.3 shall have the effect of limiting, restricting or excluding the liability of any Seller in respect of a Locked Box Claim arising as a result of such Seller's fraud.

The Buyer shall raise the Locked Box Claim no later than twelve (12) months following the Closing Date. Unless the Sellers agree otherwise in writing, if the Buyer fails to make the Locked Box Claim within the aforementioned period, it shall be deemed as the Locked Box Claim has been waived by the Buyer.

- 2.4 The Purchase Price shall be paid by the Buyer to each Seller in the following manner:
  - (a) 70% of such Seller's Purchase Price (the "**First Payment**") shall be paid on the Closing Date to the bank account designated in writing by such Seller to the Buyer no later than ten (10) Business Days prior to the Closing Date.

The remaining portion of such Seller's Purchase Price (which shall be calculated by such Seller's Purchase Price (b) minus such Seller's First Payment, the "Second Payment") shall be paid within twelve (12) months after the Closing Date to the bank account designated by such Seller.

The Parties further agree that the Buyer shall pay to each Seller the interest accrued on the Second Payment, which shall be calculated at a rate which is equal to the sum of (i) the one-year LPR applicable on the Closing Date, and (ii) 0.5%, for the period commencing from the day immediately after the Closing Date and ending on the day the Second Payment and the interest accrued thereon have been paid off. Such interest shall be paid concurrently with the Second Payment.

#### 3. CLOSING

## 3.1 Closing Date and Place

The closing of the Equity Transfer (the "Closing") shall take place remotely via the electronic exchange of the closing documents and signatures on a date mutually agreed on by the Parties but in any event no later than fifteenth (15<sup>th</sup>) Business Days following the date on which all conditions precedent set forth in Sections 4.1, 4.2 and 4.3 are satisfied or waived in writing by the relevant Party (other than conditions which by their nature are to be satisfied at the Closing), or at such other time and date as the Parties may otherwise mutually agree in writing (the date of the Closing, the "Closing Date"). The Closing with respect to each Seller shall occur concurrently.

The Parties agree that, the closing of the Lynk Capital Injection shall consummate in accordance with the terms of the Capital Injection Agreement immediately after the Closing.

#### 3.2 Closing Payment and Deliverables

- 3.2.1 At the Closing, each Seller shall deliver or cause to be delivered to the Buyer:
  - (a) with respect to GH only, a closing certificate executed by GH, dated the Closing Date, certifying that each of the conditions specified in Sections 4.1.1, and 4.2 has been satisfied; and
  - (b) with respect to VCI only, a closing certificate executed by VCI, dated the Closing Date, certifying that each of the conditions specified in Sections 4.1.2, and 4.2 has been satisfied.
- 3.2.2 At the Closing, the Buyer shall pay and deliver or cause to be paid and delivered to each Seller:
  - (a) a closing certificate executed by the Buyer, dated the Closing Date, certifying that each of the conditions specified in Section 4.3 has been satisfied; and

13

- (b) such Seller's First Payment (which may be evidenced by a copy of the receipt or payment confirmation from the bank) to the bank account designated in writing by such Seller.
- 3.2.3 At the Closing, the Company shall deliver or cause to be delivered to the Buyer:
  - the resolutions of the board of directors and the shareholders meeting of the Company approving (i) the Equity Transfer, (ii) adoption of amendment to the article of association of the Company which, among other things,
  - (a) reflects the completion of transfer of Target Equity (the "Amended Articles"), and (iii) the change of legal representative, director(s), supervisor(s) and/or general manager of the Company as set forth in Schedule VII hereto and related matters:

- (b) the Amended Articles, the updated shareholder register of the Company and the capital contribution certificate issued by the Company to the Buyer, which reflect the completion of transfer of Target Equity; and
- evidence of the completion of the AMR Registration, including the updated business license of the Company and (c) any other notification document issued by the competent AMR, which reflect the completion of transfer of Target Equity.

#### 3.3 Long Stop Date

The Parties jointly undertake to use their reasonable best efforts to procure the fulfilment of all the conditions precedents and the occurrence of the Closing as soon as possible but in any event within twelve (12) months after the Effective Date (the "Long Stop Date") or such later date as may be agreed by the Parties in writing.

3.4 Termination of Joint Venture Contract

Upon the Closing, the Amended and Restated Joint Venture Contract of the Company and its amendments executed among Ningbo Geely, VCI and GH (the "Joint Venture Contract") shall terminate automatically such that Joint Venture Contract is of no further force and effect following the Closing, and there shall be no further obligations or continuing liabilities of any of the relevant parties thereunder or in connection therewith following the Closing.

14

#### 4. CONDITIONS PRECEDENT

4.1 Conditions Precedent to Obligations of Buyer and Sellers

The obligation of the Buyer and the Sellers to consummate the transactions contemplated by this Agreement is subject to the fulfillment, on or prior to the Closing Date, of each of the following conditions (any or all of which may be waived by the Buyer and Sellers in writing in whole or in part to the extent permitted by applicable Law):

- GA's Approval. The transactions contemplated under this Agreement shall have been duly approved by the independent shareholders of GA in compliance with applicable Laws and the Rules Governing the Listing of Securities on The Stock Exchange of Hong Kong Limited.
- Volvo Cars' Approval. The transactions contemplated under this Agreement shall have been duly approved by the shareholders of Volvo Car AB (publ) in compliance with applicable Laws, the Swedish Corporate Governance Code and Nasdaq Stockholm's rulebook for issuers.
- No Injunction. There shall be no injunction, restraining order or decree of any nature of any Governmental Authority or applicable Law that is in effect that restrains or prohibits the consummation of the transactions contemplated by this Agreement.
- 4.1.4 Governmental Consent. Any applicable waiting periods, together with any extensions thereof, or any actions, non-actions, consents, approvals, waivers or clearances from any Governmental Authority that are related or required for the completion of the Equity Transfer, including those set forth in Schedule V, shall have expired, been terminated or obtained, as applicable.
- 4.2 Conditions Precedent to Obligations of the Buyer

The obligation of the Buyer to consummate the transactions contemplated by this Agreement is subject to the fulfillment, on or prior to the Closing Date, of each of the following additional conditions (any or all of which may be waived by the Buyer in writing in whole or in part to the extent permitted by applicable Laws):

4.2.1 Accuracy of Representations and Warranties. The representations and warranties of the Sellers contained in Sections 5.1 and 5.2, as applicable, shall be true and correct in all material respects, as of the date of this Agreement and as of

the Closing Date as though made on and as of the Closing Date except for representations and warranties made only at and as of a certain date, in which case as of such specified date.

- Performance of Agreement. Each of the Sellers shall have performed and complied with, in all material respects, all obligations and agreements required in this Agreement to be performed or complied with by them on or prior to the Closing Date.
- 4.2.3 No Material Adverse Effect. No Material Adverse Effect shall have occurred since the Effective Date.
- Company's Internal Approval. Each of the Sellers has delivered to the Company resolutions of the board of directors and the shareholders of the Company, duly signed by it and by the director(s) appointed by it, approving (i) implementation of the transactions contemplated by this Agreement, (ii) the adoption of the Amended Articles and (iii) the change of legal representative, director(s), supervisor(s) and/or general manager of the Company (if applicable), and evidence thereof shall have been delivered to the Buyer.

15

- 4.2.5 <u>Waiver of Shareholders' Rights</u>. The existing shareholders of the Company shall have waived their right of first refusal with respect to the Equity Transfer, and evidence thereof shall have been delivered to the Buyer.
- 4.2.6 AMR Registration. The Company shall have completed the filing and registration procedures with competent AMR in connection with (i) the Equity Transfer, (ii) the adoption of the Amended Articles, (iii) the change of legal representative, director(s), supervisor(s) and/or general manager of the Company as set forth in Schedule VII hereto (the "AMR Registration"), and evidence thereof shall have been delivered to the Buyer.
- 4.3 Conditions Precedent to Obligations of the Sellers

The obligations of the Sellers to consummate the transactions contemplated by this Agreement are subject to the fulfillment, prior to or on the Closing Date, of each of the following additional conditions (any or all of which may be waived by the Sellers in writing in whole or in part to the extent permitted by applicable Law):

- 4.3.1 Accuracy of Representations and Warranties. The representations and warranties of the Buyer contained in Section 5.1 shall be true and correct as of the date of this Agreement and as of the Closing Date as though made on and as of the Closing Date except for representations and warranties made only at and as of a certain date, in which case as of such specified date.
- 4.3.2 <u>Performance of Agreement.</u> The Buyer shall have performed and complied with, in all material respects, all obligations and agreements required in this Agreement to be performed or complied with by it on or prior to the Closing Date.

#### 5. REPRESENTATIONS AND WARRANTIES

- Each Party, severally and not jointly, hereby represents and warrants to other Parties that as of the date of this Agreement and as of the Closing Date,
  - such Party is a corporation duly and legally organized, validly existing and in good standing under relevant laws and regulations of the jurisdiction in which it is incorporated or resides;
  - such Party has all requisite power, authority and approvals required to enter into this Agreement and has all requisite power, authority and approvals to fully perform its obligations thereunder in accordance with the applicable Laws;
  - such Party's execution and performance of this Agreement do not and will not, in any material respects, (i) violate, conflict with or result in the breach of any provision of its articles of association or similar charter documents, or (ii) conflict with or violate any applicable Laws or Governmental Order applicable to it, or (iii) conflict with, result in any breach of, constitute a default (or an event which with the giving of notice or lapse of time, or both, would become a default) under, require any consent under, or give to others any rights of termination, amendment, acceleration,

- such Party's execution and performance of this Agreement do not and will not require any consent, approval, authorization or other order of, action by, filing with, or notification to, any Governmental Authority, except the Governmental Consent as set forth in Schedule V and except where in each case the failure to obtain such consent, approval, authorization, action or to make such filing or notification would not, individually or in the aggregate, prevent or materially delay the performance of such Party's obligations under this Agreement; and
- such Party's representative whose signature is affixed to this Agreement is fully authorized to sign this Agreement on behalf of such Party, and this Agreement has been duly executed and delivered by such Party and assuming the due authorization, execution and delivery by the other Parties, constitutes the legal, valid, binding and enforceable obligations of such Party in accordance with the terms hereunder.
- Each of the Sellers severally but not jointly makes the representations and warranties set out in Schedule I to the Buyer as of the date of this Agreement and as of the Closing Date. For clarity, notwithstanding anything to the contrary in this Agreement, the Sellers do not have joint and several liability for any representations and warranties, covenants and other obligations or liabilities in connection with the transaction as contemplated under this Agreement.
- The Company makes the representations and warranties set out in Schedule II to the Buyer as of the date of this Agreement and as of the Closing Date. For clarity, notwithstanding anything to the contrary in this Agreement, the Sellers do not have joint and several liability with each other and with the Company for any representations and warranties, covenants and other obligations or liabilities of the Company in connection with the transaction as contemplated under this Agreement.

#### 6. COVENANTS

- Except as otherwise expressly provided in this Agreement or with the prior written consent of the Buyer (which shall be provided promptly and not be unreasonably withheld), from the Effective Date until the earlier of the Closing Date and the valid termination of this Agreement in accordance with Section 7 (the "Interim Period"), the Company shall and shall cause its Subsidiaries to, and each Seller shall vote at the shareholders meeting of the Company to support the Company and its Subsidiaries to, to the extent permitted by applicable Laws:
  - 6.1.1 conduct their business in the ordinary course of business consistent with past practice;
  - preserve (i) the present business operations, organization (including officers and key employees) and goodwill of the Group Companies in the ordinary course of business consistent with past practice, and (ii) the present relationships with Persons having business dealings with the Group Companies (including customers and suppliers) in the ordinary course of business and consistent with past practice;

17

- maintain (i) all of the assets and properties of, or used by, the Group Companies in their current condition consistent with past practice, except for ordinary course lease expiration, and ordinary wear and tear, and (ii) insurance upon all of the properties and assets of the Group Companies in such amounts and of such kinds comparable to that in effect on the date of this Agreement;
- (i) maintain the books, accounts and records of the Group Companies in the ordinary course of business consistent with past practice, (ii) continue to collect accounts receivable and pay accounts payable using normal procedures and without discounting or accelerating payment of such accounts in the ordinary course of business consistent with past

practice, and (iii) comply with all contractual and other obligations of the Group Companies in all material respects; and

6.1.5 comply in all material respects with all applicable Laws.

6.2.5

- Without limiting the generality of Section 6.1, except as otherwise expressly provided in this Agreement or with the prior written consent of the Buyer (which shall be provided promptly and not be unreasonably withheld), during the Interim Period, the Company shall not and shall cause its Subsidiaries not to, and each Seller shall not vote at the shareholders meeting of the Company to support the Company and its Subsidiaries to, to the extent permitted by applicable Laws:
  - declare, set aside, make or pay any dividend or other distribution in respect of the share capital of, or other equity interests or ownership interests in, any Group Company or repurchase, redeem or otherwise acquire any outstanding share capital, equity interests or other securities of, or other ownership interests in, any Group Company;
  - transfer, issue, sell, pledge, encumber or dispose of any equity interests, share capital or other securities of, or other ownership interests in, any Group Company or grant options, warrants, calls or other rights to purchase or otherwise acquire equity interests, share capital or other securities of, or other ownership interests in, any Group Company;
  - effect any recapitalization, reclassification, share split, share combination or like change in the capitalization of any Group Company, or amend the terms of any outstanding securities of any Group Company;
  - amend the articles of association or equivalent organizational or governing documents of any Group Company, excluding (i) what is necessary to consummate the transactions contemplated by this Agreement, and (ii) the necessary updates to accommodate the requirement of PRC Company Law and other applicable Laws, provided that the amendments shall be made to the minimum extent required by Law and the Governmental Authorities and shall not substantially prevent or delay the consummation of the transactions contemplated by this Agreement;

18

- except for those conducted in the ordinary course of business consistent with past practice, (i) increase the salary or other compensation of any director, officer or employee of any Group Company, (ii) grant any bonus, benefit or other direct or indirect compensation to any director, officer, employee or consultant, (iii) increase the coverage or benefits available under any (or create any new) severance pay, termination pay, vacation pay, company awards, salary continuation for disability, sick leave, deferred compensation, bonus or other incentive compensation, insurance, pension or other employee benefit plan or arrangement made to, for, or with any of the directors, officers, employees, agents or representatives of any Group Company or otherwise modify or amend or terminate any such plan or arrangement or (iv) enter into or amend any employment, deferred compensation, severance, special pay, consulting, non-competition or similar agreement or arrangement with any directors, officers or employees of any Group Company;
- subject to any lien or otherwise encumber or permit, allow or suffer to be encumbered, any of the properties or assets (whether tangible or intangible, including Intellectual Properties) of, or used by, any Group Company, other than Permitted Encumbrance or other than in the ordinary course of business consistent with past practice;
- acquire any properties or assets by any Group Company with an amount in excess of RMB2,000,000 or sell, assign, 6.2.7 license, transfer, convey, lease or otherwise dispose of any of properties or assets of, or used by, any Group Company with an amount in excess of RMB2,000,000, other than in the ordinary course of business consistent with past practice;
- enter into or agree to enter into any merger or consolidation by any Group Company with any corporation or other entity, or engage any Group Company in any new business;
- 6.2.9 invest in, make a loan, advance or capital contribution to, or otherwise acquire the securities, of any other Person by any Group Company with an amount in excess of RMB2,000,000;
- 6.2.10 cancel or compromise any debt or claim or waive or release any material right of any Group Company with an amount in excess of RMB2,000,000;

- incur, assume, alter, amend or modify any indebtedness for borrowed money, or guarantee thereof, or issue any debt securities with an amount in excess of RMB2,000,000, except in the ordinary course of business consistent with past practice;
- 6.2.12 enter into any commitment for capital expenditures of any Group Company in excess of RMB2,000,000 for any individual commitment;

19

- 6.2.13 make a change in the accounting or tax reporting principles, methods or policies of any Group Company;
- enter into any Contract (i) that restrains, restricts, limits or impedes the ability of any Group Company to compete with or conduct any business or line of business in any geographic area or solicit the employment of any persons; (ii) under which the transaction amount exceeds RMB 2,000,000, except in the ordinary course of business consistent with past practice;
- enter into any Contract that would be a Material Contract if such Contract had been entered into prior to the date of 6.2.15 this Agreement or materially amend, modify, renew (other than any automatic renewal in accordance with the relevant contractual terms), consent to the termination of, or waive any material rights under, any Material Contract;
- commence any Action for a claim of more than RMB2,000,000 or settle or compromise any pending or threatened 6.2.16 Action for an amount in excess of RMB2,000,000 or that would impose any material restrictions on the business or operations of any Group Company;
- 6.2.17 effect or commence any liquidation, dissolution, winding-up, scheme of arrangement, restructuring, reorganization or similar transaction involving any Group Company;
- permit any Owned Intellectual Property of any Group Company to lapse or to be abandoned, dedicated, or disclaimed, fail to perform or make any applicable filings, recordings or other similar actions or filings, fail to pay all required fees and taxes required or advisable to maintain and protect its interest in such Owned Intellectual Property, or grant or license or transfer to any third party any such Owned Intellectual Property, except grants of non-exclusive licenses of Intellectual Property or in the ordinary course of business consistent with past practice or except where such permission, failure or grant would not result in a Material Adverse Effect;
- 6.2.19 take any action that would adversely affect the ability of the Parties to consummate the transactions contemplated by this Agreement; or
- agree to do anything (i) prohibited by this Section 6.2, (ii) that would make any of the representations and warranties of the Sellers in this Agreement untrue or incorrect in any material respect or could result in any of the conditions to the Closing not being satisfied or (iii) that would be reasonably expected to have a Material Adverse Effect.

#### 6.3 Access to Information

In each case subject to restrictions under applicable Laws and the Group Companies confidentiality obligations toward third parties,

as soon as reasonably practicable, but in no event later than twenty (20) days after the end of each calendar month during the Interim Period, the Company shall, and each Seller shall vote at the shareholders meeting of the Company to support to, provide the Buyer with (i) unaudited monthly financial statements and (ii) operating or management information of the Group Companies reasonably required by the Buyer. In addition, the Company shall and each Seller shall vote at the shareholders meeting of the Company to support to, promptly provide the Buyer with other information reasonably requested by the Buyer, including any information requested in connection with the Buyer's financing for the Equity Transfer.

the Company and its Subsidiaries shall allow the Buyer and its accountants, counsel, financial advisors and other representatives, at their own cost, reasonable access, during normal business hours upon prior notice during the Interim Period, to the Group Companies' respective properties and facilities (including all real property and the buildings, structures, fixtures, appurtenances and improvements erected, attached or located thereon), books, financial information, contracts and records of the Group Companies and, during such period, shall as soon as reasonably possible furnish such information concerning the businesses, properties and personnel of the Group Companies as the Buyer shall reasonably request; provided, however, that such access in each case shall not disrupt the Group Companies' operations.

## 6.4 Notification of Certain Matters

Each of the Parties shall give notice to the other Parties as promptly as reasonably practicable upon becoming aware of (a) any fact, change, condition, circumstance, event, occurrence or non-occurrence that has caused or is reasonably likely to cause any representation or warranty in this Agreement made by it to be untrue or inaccurate in any respect at any time after the date hereof and prior to the Closing, (b) any failure on its part to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it hereunder, (c) the institution of or the threat of institution of any Action against it that may adversely affect the transactions contemplated hereby or the Group Companies or related to this Agreement or the transactions contemplated hereby, (d) any notice or other communication from any Governmental Authority in connection with the transactions contemplated by this Agreement, or (e) any notice or other communication from any Person alleging that the consent of such Person is or may be required in connection with the transactions contemplated by this Agreement; provided that the delivery of any notice pursuant to this Section 6.4 shall not limit or otherwise affect the remedies available hereunder to the Party receiving such notice, or the representations or warranties of, or the conditions to the obligations of, the Parties hereto.

#### 6.5 Further Assurances

Each of the Parties shall use its reasonable best efforts to take, or cause to be taken, all actions necessary or appropriate to (i) consummate the transactions contemplated by this Agreement and (ii) cause the fulfillment at the earliest practicable date of all of the conditions to their respective obligations to consummate the transactions contemplated by this Agreement. Without limiting the generality of the foregoing, prior to and at the Closing Date, each Party shall cooperate with the other Parties without any further consideration to make all filings with, and to obtain all consents of, any Governmental Authority or any other Person under any permit, license, agreement, indenture or other instrument (including any consents), and to take all such other actions as such Party may reasonably be requested to take by the other Parties from time to time, consistent with the terms of this Agreement, in order to effectuate the provisions and purposes of this Agreement and the transactions contemplated hereby.

21

#### 6.6 GH's Additional Covenants

GH shall use reasonable best efforts to assist GA (i) to convene a shareholders meeting as soon as possible to authorize and approve the transactions contemplated in this Agreement, (ii) to recommend to its shareholders that they authorize and approve this Agreement and the transactions contemplated hereunder, (iii) to solicit from its shareholders proxies in favor of the authorization and approval of this Agreement and the transactions contemplated hereunder, and (iv) to take all other actions necessary or advisable to secure the approval from its independent shareholders, in each case in compliance with applicable Laws and the Rules Governing the Listing of Securities on The Stock Exchange of Hong Kong Limited.

6.6.2

# 6.7 VCI's Additional Covenants

VCI shall use reasonable best efforts to cause Volvo Car AB (publ) (i) to convene a shareholders meeting as soon as reasonably possible (considering e.g. timing for GH's shareholders meeting) to authorize and approve the transactions contemplated in this Agreement, (ii) to recommend to its shareholders that they authorize and approve this Agreement and the transactions contemplated hereunder, and (iii) to take all other actions reasonably necessary or advisable to secure shareholders' approval, in each case in compliance with applicable Laws, the Swedish Corporate Governance Code and Nasdaq Stockholm's rulebook for issuers.

# 6.8 Company's Additional Covenants

6.8.1

22

6.8.2

6.8.3

(e)

# 7. TERMINATION

- 7.1 This Agreement may be terminated at any time prior to the Closing:
  - (a) by the mutual written consent of the Parties;
  - by any of the Sellers or the Buyer, by written notice to the other Parties, if the Closing has not occurred on or before the Long Stop Date, provided that the right to terminate this Agreement pursuant to this Section 7.1(b) shall not be available to the Seller or the Buyer seeking to terminate if such Seller or the Buyer is in breach of, or has breached, any of its obligations under this Agreement, which breach has been a material cause of the failure of the Closing to be consummated on or before such date;
  - by any Party, by written notice to the other Parties, if the consummation of the transactions contemplated by this Agreement would violate any non-appealable final Governmental Order, provided that the right to terminate this (c) Agreement pursuant to this Section 7.1(c) shall not be available to the Party seeking to terminate if such Party is in breach of, or has breached, any of its obligations under this Agreement, which breach has been a material cause of such Governmental Order;
  - by written notice from the Buyer if there has been a material breach of any representation, warranty, covenant or agreement on the part of any Seller contained in this Agreement such that the conditions set forth in Section 4.2 would not be satisfied and such breach or failure of condition is not curable or, if curable, is not cured prior to the earlier of (i) 30 days after written notice thereof is given by the Buyer to such Seller or (ii) the Long Stop Date; provided that the Buyer shall not have the right to terminate this Agreement pursuant to this Section 7.1(d) if the Buyer is then in breach of any of its representation, warranty, covenant or agreement contained in this Agreement that would cause the conditions set forth in Section 4.3 not to be satisfied; or

23

by written notice from any Seller if there has been a material breach of any representation, warranty, covenant or agreement on the part of the Buyer contained in this Agreement such that the conditions set forth in Section 4.3 would not be satisfied and such breach or failure of condition is not curable or, if curable, is not cured prior to the earlier of (i) 30 days after written notice thereof is given by such Seller to the Buyer or (ii) the Long Stop Date; provided that such Seller shall not have the right to terminate this Agreement pursuant to this Section 7.1(e) if such Seller is then in breach of any of its representation, warranty, covenant or agreement contained in this Agreement that would cause the conditions set forth in Section 4.2 not to be satisfied.

In the event of termination by any Party pursuant to Section 7.1 above, written notice thereof shall promptly be given to the other Parties and the transaction contemplated hereunder shall be terminated without further action by any Party; provided, however, that (a) the provisions of this Section 7 (*Termination*), Section 9 (*Confidentiality*), Section 10 (*Notice*), Section 11 (Governing *Law and Dispute Resolution*), Section 12 (*Taxes, Costs and Expenses*) and Section 13 (*Miscellaneous*) shall survive such termination, (b) the Parties shall take all necessary actions (including without limitation to cancel or modify any government registration, filings or other documents that have been conducted before such termination) to restore all Parties to their original position and status they were in immediately prior to the execution of this Agreement and the Party which is in breach of the Agreement (if any) shall be responsible for the relevant fees and Losses, and (c) nothing contained herein shall relieve any Liabilities of any Party for breach of this Agreement that occurred prior to the termination hereof.

# 8. INDEMNIFICATION AND REMEDIES

# 8.1 Survival

7.2

All representations and warranties made by each Party contained in Section 5 shall survive for a period of twenty-four (24) months following the Closing Date; provided that (i) the Fundamental Representations shall survive for a period of thirty-six (36) months following the Closing Date and (ii) the representations and warranties set forth in Section 8 of Schedule II shall survive until 60 days after the expiration of the applicable statute of limitations for the taxes at issue. The covenants and agreements of each Seller and the Company set forth in this Agreement shall survive the Closing until fully discharged in accordance with their terms and the covenants or agreements that by their terms relates only to the Interim Period shall terminate as of the Closing.

## 8.2 Indemnification by the Company and Sellers

Subject to the limitations as set forth in Section 8 (including Section 8.2.4 and Section 8.2.6), from and after the Closing, each Seller and the Company (each, an "Indemnifying Party") hereby agrees to, severally but not jointly, indemnify and hold the Buyer, its Affiliates and their respective directors, officers, employees, agents, successors and assignees (collectively, the "Indemnified Parties" and each, an "Indemnified Party") harmless from and against any and all Losses which arise out of or result from: (i) any material breach, violation or nonperformance of any covenant or agreement of such Indemnifying Party under this Agreement, and/or (ii) any material inaccuracy or breach of any of the representations or warranties made by such Indemnifying Party under this Agreement. For the avoidance of doubt, nothing in this Clause released the Indemnified Parties to prove their Losses.

24

- For purposes of applying the indemnification remedies provided in this Section 8, when calculating the amount of any Losses relating thereto, in each case, the representations, warranties and covenants made by any Indemnifying Party in this Agreement shall be considered and applied without regard to any reference as to materiality, Material Adverse Effect or similar materiality qualifications set forth therein.
- 8.2.3 Notwithstanding anything to the contrary herein, each Seller's and the Company's liability towards the Buyer under this Agreement is subject to Closing having occurred.

# 8.2.4 Limitations in Amount

Absent fraud and willful breach, the Indemnified Party shall not be entitled to indemnification under Section 8.2.1 unless, with respect to the claims of the Indemnified Party, (i) the amount of the relevant single claim thereunder exceeds RMB 1,000,000 (each a "Qualifying Claim"), and (ii) the aggregate amount of all Qualifying Claims thereunder exceeds RMB 45,000,000 (the "Basket Amount"), in which case the Indemnified Party shall be entitled to the full amount of such claim(s) for the entire amount of such Losses, and not merely the portion of such Losses exceeding the Basket Amount, subject to the Seller Cap and Company Cap defined below;

- Absent fraud and willful breach, the total liability of each Seller in respect of all Loss(es) under Section 8.2.1 shall (b) not exceed 10% of such Indemnifying Party's Purchase Price that has been received by the Indemnifying Party (the "Seller Cap"); and
- (c) Absent fraud and wilful breach, the total liability of the Company in respect of all Loss(es) under Section 8.2.1 shall not exceed 10% of the total Purchase Price payable by the Buyer ("Company Cap").
- Notwithstanding any other provisions of this Agreement to the contrary but without prejudice to Section 8.4, for any indemnification claim made by an Indemnified Party against any Indemnifying Party under this Section 8, if the Indemnifying Party is the Company, the Indemnifying Party shall not have the right of recourse against the Sellers in any event, and if the Indemnifying Party is the Sellers, the Indemnifying Party shall not have the right of recourse against any of the Group Companies in any event.

25

- 8.2.6 The Indemnifying Party shall not be liable to compensate the Indemnified Party for any Loss:
  - if the matter or occurrence giving rise to the breach of representations and warranties and the subsequent Loss has been included in the Locked Box Account or has been otherwise disclosed to the Buyer, including but not limited to those disclosed by the Sellers or the Group Companies in the Disclosure Letter (including the general disclosure and specific disclosure thereunder), the Financial Statements or this Agreement;
  - (b) which occurs as a result of any act or omission by the Indemnifying Party in accordance with any request or consent from the Indemnified Party; and
    - which occurs as a result of any applicable Laws not issued on the Effective Date (or any alteration or repeal of any applicable laws in force on the Effective Date, to the extent that such alteration or repeal has not been issued on
  - (c) the Effective Date), or which takes effect retroactively, or occurs as a result of any increase in the tax rate in force on the Effective Date or any change in the generally established practices (including interpretation of applicable Laws) of the relevant authority.
- Notwithstanding anything to the contrary herein, the Buyer acknowledges that it has had the opportunity to conduct due diligence and investigation with respect to the Group Companies, and in no event shall the Sellers have any liability to the Buyer with respect to a breach of representation, warranty or covenant under this Agreement to the extent that the Buyer knew or should have known (after making such due inquiry and exercising such due diligence as a prudent investor would have made or exercised in connection with a comparable investment) of such breach as of the Closing Date.

# 8.3 Indemnification by Buyer

- If any sum due for payment in accordance with this Agreement is not paid by the Buyer on the due date for payment, the Buyer shall pay a late payment interest at the rate of 0.01% per day on that sum from but excluding the due date to and including the date of actual payment calculated on a daily basis.
  - Subject to the limitations as set forth in Section 8.3.3 and Section 8.4, from and after the Closing, the Buyer hereby agrees to indemnify and hold the Sellers, its Affiliates and their respective directors, officers, employees, agents, successors and assignees (collectively, the "Seller Indemnified Parties" and each, a "Seller Indemnified Party")
- 8.3.2 harmless from and against any and all Losses which arise out of or result from: (i) any material breach, violation or nonperformance of any covenant or agreement of the Buyer under this Agreement, and/or (ii) any material inaccuracy or breach of any of the representations or warranties made by the Buyer under this Agreement. For the avoidance of doubt, nothing in this Clause released the Seller Indemnified Parties to prove their Losses.
- Absent fraud and willful breach, each Seller Indemnified Party shall not be entitled to indemnification under this Section 8.3.2 unless, with respect to the claims of the Seller Indemnified Party, (i) the amount of the relevant single claim thereunder exceeds RMB 1,000,000 (each a "Seller Qualifying Claim"), and (ii) the aggregate amount of all

- Absent fraud and willful breach, the total liability of the Buyer in respect of all Loss(es) under Section 8.3.2 of each Seller Indemnified Party shall not exceed 10% of such Seller's Purchase Price that shall be payable by the Buyer (the "Buyer Cap").
- 8.3.5 Notwithstanding anything to the contrary herein, the Buyer's liability towards the Sellers under this Agreement is subject to Closing having occurred.
- Notwithstanding anything to the contrary herein, the Buyer's obligations of paying the Purchase Prices to the Sellers under Section 2.2, Section 2.4, the interest accrued on the Second Payment and the interest under Section 8.3.1 above are not subject to any liability limitations or other restrictions as provided in Sections 8.3.3 and 8.3.4, and none of Sections 8.3.3 or 8.3.4 shall apply to the obligation and liabilities in connection with Buyer's payment of the Purchase Prices or any interest accrued thereon to the Sellers under Section 2.2, Section 2.4 and Section 8.3.1.
- Any Losses subject to indemnification hereunder shall be determined without duplication of recovery by reason of the state of facts giving rise to such Losses, including if such state of facts constitutes a breach of more than one representation, warranty, covenant or other provision of this Agreement.

#### 9. CONFIDENTIALITY

9.1

9.2

Subject to Sections 9.2 and 9.3, each Party undertakes with the other Parties that it shall treat as strictly confidential all information received or obtained by it, its Affiliates or the directors, officers, employees, agents, advisers or other representatives of such Party or its Affiliates (collectively, such Party's "Representatives") in connection with entering into or performing this Agreement, including information relating to the existence and provisions of this Agreement, the negotiations leading up to this Agreement, the materials and information acquired by the Representatives through their due diligence of the Group Company, the subject matter of this Agreement or the business or affairs of the other Parties or any Subsidiary of the other Parties (collectively, "Confidential Information"), and that it shall not at any time hereafter disclose or divulge to any Person any Confidential Information and shall use its best endeavor to prevent the publication or disclosure of any Confidential Information. Each Party also undertakes with the other Parties that neither it nor any of its Representatives shall trade or engage in any derivative or other transaction on the basis of such Confidential Information in violation of any applicable Laws.

Confidential Information shall not include any information that is (i) previously known on a non-confidential basis by the receiving Party or its Representatives, (ii) in the public domain through no fault of such receiving Party or its Representatives, (iii) received from a Person other than any of the other Parties or their respective Representatives, so long as such Person was not, to the best knowledge of the receiving Party, subject to a duty of confidentiality to such other Party, or (iv) developed independently by the receiving Party without reference to Confidential Information of the disclosing Party.

27

# 9.3 Notwithstanding Section 9.1:

any Party may disclose Confidential Information to the extent that such disclosure (i) is required under applicable Laws or any judicial or regulatory process or (ii) is requested by any Governmental Authority or other regulatory body, including the rules and requirements of any securities exchange; or (iii) is made in relation to the arbitration as set forth in Section 11.2, provided, that, in the case of the foregoing (i) and (ii), such Party shall, to the extent permitted by Law and so far as it is practicable, provide the other Parties with prompt notice of such requirement or request and

cooperate with the other Parties at such other Parties' request and cost to enable such other Parties to seek an appropriate protection order or remedy; and

any Party may disclose Confidential Information to its Representatives on a need-to-know basis; provided, that such Party shall use reasonable best efforts to ensure that each such Person to which it discloses Confidential Information strictly abides by the confidentiality obligations hereunder and shall be responsible for any breach of confidentiality obligations by such Person.

Except as may be required by applicable Law, the press release announcing the execution of this Agreement shall be issued only in such form as shall be agreed among the Parties. Thereafter, at any time prior to the earlier of the Closing Date and the valid termination of this Agreement pursuant to Section 7, except as may be required by applicable Law, each Party shall consult with the other Parties before it or its Affiliates issues any press release, has any communication with the press, making any other public statement with respect to this Agreement and the transactions contemplated hereunder, and shall provide each other a reasonable opportunity to review and comment on (and consider such proposed comments in good faith), such press releases, communication or public statement.

## 10. NOTICE

9.4

10.1

Any notice given by a Party to the other Parties under this Agreement shall be deemed validly served by hand delivery or by prepaid registered mail sent through the post (airmail if to an overseas address) or by email transmission to its address given herein or such other address as may from time to time be notified for this purpose and any notice served by prepaid registered letter shall be deemed to have been served 48 hours (72 hours in the case of a letter sent by airmail to an address in another country) after the time at which it was posted, any notice served by email transmission shall be deemed to have been served at the time of the email transmission, it shall be sufficient (in the case of service by hand and prepaid registered letter) to prove that the notice was properly addressed and delivered or posted, as the case may be, and in the case of service by email transmission to prove that there is no system-generated non-delivery notice received by the sender.

28

Each notice, demand or other communication given or made under this Agreement shall be in writing and delivered or sent to the relevant Party at its address or telex number or email address set out hereinafter (or such other address or email address as the addressee has by five (5) days' prior written notice specified to the other Parties):

To the Buyer

:

To GH :

To VCI :

To the Company

#### 11. GOVERNING LAW AND DISPUTE RESOLUTION

- The validity, construction, implementation and interpretation of this Agreement shall be governed by and construed in accordance with the laws of the PRC.
- 11.2 Dispute resolution
  - 11.2.1 Consultations

In the event of a dispute arising out of or in connection with the interpretation or implementation of this Agreement (the "**Dispute**"), the Parties shall attempt in the first instance to resolve the Dispute in good faith by friendly consultations between the Parties. If the Dispute cannot be resolved by any such method within thirty (30) Business Days after the commencement of consultation, then any Party may submit the Dispute to arbitration.

#### 11.2.2 Arbitration

- Any Dispute referred to arbitration shall be finally resolved by arbitration administered by the China International Economic and Trade Arbitration Commission Shanghai Sub-Commission ("CIETAC"), which shall be held in Shanghai and conducted in accordance with the CIETAC arbitration rules in force when the notice of arbitration is submitted by a Party, which rules are deemed to be incorporated by reference into this clause, by three (3) arbitrators appointed in accordance with the said rules, and in such case:
  - (i) all proceedings in any such arbitration shall be conducted in English;
  - (ii) all of the arbitrators shall be fluent in English; and
  - (iii) only the English text of this Agreement shall be referred to by the arbitrators in the course of the arbitration proceedings.
- In any arbitration proceeding, any legal proceeding to enforce any arbitration award and in any legal action between the Parties pursuant to or relating to this Agreement, each Party expressly waives the defence of sovereign immunity and any other defence based on the fact or allegation that it is a political subdivision, agency or instrumentality of a sovereign state.
- Should a Dispute be submitted to arbitration, all Parties shall in all other respects, except in the event of termination, continue to perform their respective obligations in accordance with this Agreement.
- The arbitration award shall be final and binding on all Parties. The costs of arbitration shall be borne by the losing Party or as otherwise determined by the arbitration tribunal. Any award of the arbitrator tribunal shall be enforceable by any court having jurisdiction over a Party against which the award has been rendered, or in any place where assets of a Party against which the award has been rendered are located, and will be enforceable in accordance with applicable Law in the jurisdiction in which such enforcement is sought, and the terms of applicable treaties and international conventions.

#### 12. TAXES, COSTS AND EXPENSES

Unless as otherwise expressly specified by this Agreement, each Party shall bear its own taxes, costs and expenses incurred in connection with negotiation, preparation, execution and performance of this Agreement, including but not limited to, stamp duty, income tax, financial and legal due diligence costs, valuation fees or other legal fees.

## 13. MISCELLANEOUS

Each Party hereby agrees, represents and warrants to the other Parties that such Party (i) is not, and is not acting for or on behalf of, a Listed Person, and (ii) shall not, in the course of performing under this Agreement (a) engage in any transaction that evades, avoids or attempts to violate any Sanctions which the other Parties or their Affiliates shall comply with, or (b) pay to the other Parties any proceeds derived from a transaction with a Listed Person or a Sanctioned Country.

- Neither Party shall be entitled to a duplicative compensation for the same Loss under this Agreement and the Capital Injection Agreement.

  This Agreement shall take effect upon signing by the Parties on the date first written above and shall be binding upon and ensure for the benefit of the estates, successors and/or permitted assigns of the Parties.

  Each of the Parties hereto undertakes to the other Parties that it shall do all such acts and things and execute all such deeds and documents as may be necessary or desirable to carry into effect or to give legal effect to the provisions of this Agreement and
  - the transactions hereby contemplated.
- This Agreement constitutes the whole agreement between the Parties and supersede and terminate any previous agreements or arrangements between them relating to the subject matter hereunder; it is expressly declared that no variations hereof shall be effective unless made in writing and signed by duly authorized representatives of the Parties.
- If any provision or part of a provision of this Agreement shall be, or be found by any authority or court of competent jurisdiction to be, invalid or unenforceable, such invalidity or unenforceability shall not affect the other provisions or parts of such provisions of this Agreement, all of which shall remain in full force and effect.
- The Buyer and the Sellers may separately sign an equity transfer agreement in simple form ("**Short Form Agreement**"), which shall be solely used for filing and registration with competent Governmental Authority for completion of relevant administrative procedures with respect to the Equity Transfer. For the avoidance of doubt, this Agreement shall prevail in case that the Short Form Agreement conflicts with this Agreement.
- The Parties to this Agreement are independent of each other. Each Party shall exercise its rights and assume its obligations hereunder independently. Any act or judgment by either Party shall neither affect the other Parties nor affect the rights and obligations of the other Parties hereunder or the force of any clause hereof concerning the other Parties. For further clarification, the obligations of each Seller and the Group Companies under this Agreement are several and not joint, and (i) neither Seller shall be liable or responsible in any way for the actions or lack of actions of the other Seller in connection with the transaction as contemplated under this Agreement; (ii) neither Seller shall be liable or responsible in any way for the actions or lack of actions of the Group Companies in connection with the transactions as contemplated under this Agreement.
- No Party may assign this Agreement and/or any of its rights, interests, or obligations hereunder without the prior written approval of the other Parties.
- This Agreement may be executed in one or more counterparts, and by the Parties on separate counterparts, but shall not be effective until each Party has executed at least one counterpart and each such counterpart shall constitute an original of this Agreement but all the counterparts shall together constitute one and the same instrument.

31

This Agreement is made in English. If a Chinese translation of this Agreement is delivered it is for information purposes only and does not constitute a definitive and legally binding agreement. For the avoidance of doubt, in case there is any conflict between the Chinese translation and the English version, the English version shall prevail.

< Remainder of this page intentionally left blank >

32

Signature page to the EQUITY TRANSFER AGREEMENT

On behalf of
Zhejiang ZEEKR Intelligent Technology Co., Ltd. (浙江极氪智能科技有限公司)
(Company seal)
Signature:/s/ Conghui An Name: Conghui An Title: CEO
Signature page to the EQUITY TRANSFER AGREEMENT
IN WITNESS WHEREOF the Parties have hereunto caused this Agreement to be duly executed as of the date first above written.
On behalf of
Zhejiang Geely Holding Group Co., Ltd. (浙江吉利控股集团有限公司)
(Company seal)
Signature:/s/ Donghui Li Name: Donghui Li Title: Legal Representative
Signature page to the EQUITY TRANSFER  AGREEMENT
<b>IN WITNESS WHEREOF</b> the Parties have hereunto caused this Agreement to be duly executed as of the day and year first above written.
On behalf of
Volvo Cars (China) Investment Co., Ltd. (沃尔沃汽车(中国)投资有限公司)
(Company seal)
Signature:/s/ Xiaolin Yuan Name: Xiaolin Yuan

Legal Representative

Title:

Signature page to the EQUITY TRANSFER

AGREEMENT

**IN WITNESS WHEREOF** the Parties have hereunto caused this Agreement to be duly executed as of the date first above written.

On behalf of

LYNK & CO Automotive Technology Co., Ltd. (领克汽车科技有限公司)

(Company seal)

Signature:/s/ Gan Jiayue

Name: Gan Jiayue

Title: Legal Representative / Director

# Schedule I Representations and Warranties of the Sellers

As an inducement for the Buyer to enter into this Agreement, each Seller as the existing shareholder of the Company, severally but not jointly, to the knowledge of the Sellers, makes the following representations and warranties to the Buyer as of the Effective Date and as of the Closing Date:

# 1. Organization and Good Standing

The Company is duly organized, validly existing and in good standing (or equivalent status in the relevant jurisdiction) under the Laws of the place of its incorporation or establishment.

## 2. Capitalization

The Target Equity held by such Seller, when transferred to the Buyer in accordance with the terms and conditions of this Agreement, is free and clear of any Encumbrances (except for the Encumbrances which have been disclosed to the Buyer, the pre-emptive right and right of first refusal specified by applicable Laws and the Company's articles of associations or similar charter document, if any).

# 3. Ownership of the Target Equity

As of the Effective Date and immediately prior to the Closing, such Seller is the record and beneficial owner and holder of the Target Equity held by it, fully paid up, validly issued and free and clear of any and all Encumbrance (except for the pre-emptive right and right of first refusal specified by applicable Laws and the Company's articles of associations or similar charter document and the Encumbrances which have been disclosed to the Buyer, if any). Such Seller has the power and authority to sell, transfer, assign and deliver the Target Equity as provided in this Agreement. Such Seller obtained the Target Equity in accordance with applicable Law, and none of the funds of such Seller used to purchase the Target Equity were proceeds obtained directly by such Seller as a result of any activity in violation of applicable Law that could, as a result of such violation, prejudice such Seller's title over the Target Equity. Upon the transfer of the Target Equity to the Buyer on the Closing Date in accordance with this Agreement, the Buyer will receive good and valid title to the Target Equity and all rights attached to the Target Equity, free and clear of all Encumbrances (except for the pre-emptive right and right of

first refusal specified by applicable Laws and the Company's articles of associations or similar charter document and the Encumbrances which have been disclosed to the Buyer, if any).

i

# Schedule II Representations and Warranties of the Company

As an inducement for the Buyer to enter into this Agreement, the Company makes the following representations and warranties to the Buyer as of the Effective Date and as of the Closing Date, except as set forth in, or qualified by any matter set forth in the disclosure letter as attached hereto as Schedule III (the "**Disclosure Letter**"). For clarity, none of the representations and warranties shall be deemed as having been made by any Seller in either an express or an implied way.

## 1. Organization and Good Standing

Each of the Group Companies is duly organized, validly existing and in good standing (or equivalent status in the relevant jurisdiction) under the Laws of the place of its incorporation or establishment and has all necessary corporate power and authority to own, operate or lease all necessary properties and assets and to conduct its material businesses as presently used and conducted. Each of the Group Companies is duly qualified or licensed to conduct its business in all jurisdictions in which it conducts its business, and such qualification or license is in force and effect, except where the failure to be so qualified or licensed would not cause a Material Adverse Effect on the Group Companies, taken as a whole. All articles of association or equivalent organizational documents of the Group Companies are in full force and effect. All material corporate actions taken by the Group Companies have been duly authorized, and the Group Companies have conducted their business in all material respects in compliance with their respective articles of association or equivalent organizational documents. None of the Group Companies has winded up its business or become bankrupt or insolvent, or lost its ability to repay its due debt, nor has it entered into any liquidation of bankruptcy proceeding pursuant to the applicable Laws. No application made, order made, valid resolution passed or other actions have ever been made, issued, adopted or taken with respect to its liquidation, winding-up, bankruptcy announcement or other similar events.

## 2. Capitalization

- (a) The registered capital of the Company has been fully paid up in compliance with all requirements of the applicable Laws and the articles of association of the Company. The Target Equity, when transferred to the Buyer in accordance with the terms and conditions of this Agreement, is free and clear of any Encumbrances (except for the pre-emptive right and right of first refusal specified by the applicable Laws, the Amended Articles and similar charter documents of the Company).
- (b) Except as set forth in Schedule VI and the Lynk Capital Injection, (i) there is no issued, reserved for issuance or outstanding capital stock or other voting securities of or ownership interests in the Company, (ii) securities of the Company convertible into or exchangeable for shares of capital stock or other voting securities of or ownership interests in the Company, (iii) options, warrants, convertible securities or other rights, agreements, arrangements or covenants of whatever nature pertaining to the capital stock or other voting securities or ownership interests in the Company, or obligating the Company to sell any of its capital stock or other voting securities or ownership interests, or (iv) other securities or rights that are derivative of, or provide economic benefits based, directly or indirectly, on the value or price of, any capital stock or other voting securities of or ownership interests in the Company (the items in clauses (i) through (iv) collectively, the "Company Securities"). There are no outstanding obligations of the Company or other Group Company to repurchase, redeem or otherwise acquire any of the Company Securities. The Company does not have outstanding any bonds, debentures or notes that provide the holders thereof with the right to vote (or are convertible into or exchangeable or exercisable for securities having the right to vote) on any matter on which the shareholders of the Company may vote.

ii

(c) There are no pending disputes with respect to equity interest of any Group Company. There is no nominee holding of equity interest.

## 3. Subsidiaries

- (a) Schedule VI sets forth a true, correct and complete list of the Group Companies and such other Person in which the Company or any other Group Company owns any equity interests. Except as set forth in Schedule VI, no Group Company owns or controls any equity interest in any other Person, and no Group Company is obligated to make any investment in or capital contribution in or on behalf of any other Person.
- (b) All of the outstanding capital stock or other voting securities of or ownership interests in, each Subsidiary of the Company, are owned by the Company, directly or indirectly, free and clear of any Encumbrances (except for the pre-emptive right and right of first refusal specified by the applicable Laws and their articles of associations or similar charter documents). Except as set forth in Schedule VI, there is no (i) issued, reserved for issuance or outstanding capital stock or other voting securities of or ownership interests in any Subsidiary of the Company, (ii) shares of capital stock or other voting securities of or ownership interests in any Subsidiary of the Company, (iii) securities of the Company or any of its Subsidiaries convertible into or exchangeable for shares of capital stock or other voting securities of or ownership interests in any Subsidiary of the Company, (iv) options, warrants, convertible securities or other rights, agreements, arrangements or covenants of whatever nature pertaining to the capital stock or other voting securities or ownership interests in any Subsidiary, or obligating any Subsidiary of the Company to sell any of its capital stock or other voting securities or ownership interests, or (v) other securities or rights that are derivative of, or provide economic benefits based, directly or indirectly, on the value or price of, any capital stock or other voting securities of or ownership interests in any Subsidiary of the Company (the items in clauses (i) through (v) collectively, the "Company Subsidiary Securities"). There are no outstanding obligations of the Company or any of its Subsidiaries to repurchase, redeem or otherwise acquire any of the Company Subsidiary Securities.
- (c) Except for the non-payment of the registered capital that is in compliance with the applicable Laws and their articles of associations, the outstanding shares, share capital or registered capital, as the case may be, of each of the Subsidiaries of the Company is duly authorized, validly issued, fully paid and nonassessable, and the portion of the outstanding shares, share capital or registered capital, as the case may be, of each of the Company's Subsidiaries and such other entity in which a Group Company owns or otherwise holds any equity interest is owned by such Group Company free and clear of all Encumbrances (except for the pre-emptive right and right of first refusal specified by applicable Laws and their articles of association or similar charter documents).

iii

## 4. Ownership of the Target Equity

As of the Effective Date and immediately prior to the Closing, (i) GH is the record and beneficial owner and holder of GH Equity, fully paid up, validly issued and free and clear of any and all Encumbrance (except for the pre-emptive right and right of first refusal specified by applicable Laws and the Company's articles of associations or similar charter documents) and (ii) VCI is the record and beneficial owner and holder of VCI Equity, fully paid up, validly issued and free and clear of any and all Encumbrance (except for the pre-emptive right and right of first refusal specified by applicable Laws and the Company's articles of associations or similar charter documents). Upon the transfer of the Target Equity to the Buyer on the Closing Date in accordance with this Agreement, the Buyer will receive good and valid title to the Target Equity and all rights attached to the Target Equity, free and clear of all Encumbrances (except for the pre-emptive right and right of first refusal specified by applicable Laws and the Company's articles of associations or similar charter documents).

# 5. Financial Statements

- (a) The Company has delivered to the Buyer copies of the consolidated balance sheets of the Group Companies as at December 31, 2021, December 31, 2022, December 31, 2023 and September 30, 2024 and the related consolidated statements of income and of cash flows of the Group Companies for the years/periods then ended (such statements, including the related notes and schedules thereto, are referred to herein as the "Financial Statements"). The Financial Statements and the Locked Box Accounts: (i) have been prepared according to the books and other financial records of the Group Companies, (ii) in all respects, present truly, accurately, completely and fairly the financial conditions, results of operations and cash flows of the Group Companies, as of the dates thereof (or with respect to the Locked Box Accounts, as of the Locked Box Date) or for the periods covered thereby, and (iii) have been prepared in accordance with the Accounting Standards and the accounting principles of the Group Companies on a consistent basis.
- (b) The books and other financial records of the Group Companies: (i) reflect in all material aspects all items of income and expense, and all assets and Liabilities required to be reflected therein in accordance with the Accounting Standards, (ii) are complete and accurate in

all material respects, and do not contain or reflect any material inaccuracies or discrepancies, and (iii) have been prepared in accordance with good business and accounting practices in all material aspects.

- (c) The assets and inventories of the Group Companies are in good saleable condition and to the knowledge of the Company, no impairment is needed.
- (d) There has not been any Leakage from (but excluding) the Locked Box Date to (and including) the date of this Agreement.

## 6. Absence of Certain Changes

Since the Locked Box Date, there has not been any Material Adverse Effect and there has not been any fact, event, change or circumstance that would be reasonably likely to have a Material Adverse Effect. Except as a result of the execution and delivery of this Agreement, since the Locked Box Date, (i) the business of the Group Companies has been conducted in all material respects in the ordinary course consistent with past practice and (ii) no Group Company has taken any action or omitted to take any action that would, if taken after the date hereof, be prohibited by Section 6.2.

iv

#### 7. No Undisclosed Liabilities.

No Group Company has any Liabilities, except for (i) Liabilities reflected in the Financial Statements or the Locked Box Accounts, (ii) Liabilities incurred in the ordinary course of business consistent with past practice since the Locked Box Date that would not be prohibited by this Agreement, or (iii) Liabilities that would not materially and adversely impact the Group Companies, taken as a whole.

#### 8. Taxation

- (a) Except as would not reasonably be expected to have a Material Adverse Effect, each Group Company, within applicable time limits, (i) has filed all material tax returns required to be filed under applicable Laws (or has been properly included in all tax returns required to be filed under applicable Laws), and such tax returns are true, correct, and complete without any omission in any material respects; (ii) has obtained and maintained the invoices (including VAT invoices) receipts and other supporting documents that comply with applicable Laws in all material respects; (iii) has provided all information and maintained in all material respects all records, receipts and other supporting documents in relation to tax as it is required to make, provide or maintain; and (iv) has complied in all material respects on a timely basis with all notices served on it and any other requirements lawfully made of it by any tax authority. Except as would not reasonably be expected to have a Material Adverse Effect, to the knowledge of the Company, each Group Company has timely paid or withheld all taxes due, and it is or will not become liable to pay any fine, penalty, surcharge or interest in relation to tax with respect to any taxable period prior to the date of this Agreement.
- (b) Each Group Company is in compliance in all material respects with all applicable transfer pricing requirements in all jurisdictions in which it does business.
- (c) None of the Group Companies has unsettled Actions that may have Material Adverse Effect with any tax authorities.

# 9. Assets

The Group Companies have in all material respects good and marketable title to, or a valid and binding leasehold interest in, all material personal properties and assets reflected on the Financial Statements and all other properties and assets necessary to conduct their respective businesses, in each case free and clear of all Encumbrance of any kind except for Permitted Encumbrances. With respect to the material property and assets it leases, each of the Group Companies is in all material respects in compliance with such leases

# 10. Real Property

(a) Each Group Company has good and valid title to its Owned Real Property, free and clear of all Encumbrances except for Permitted Encumbrances. Except as would not reasonably be expected to have a Material Adverse Effect, each Group Company has duly obtained for each Owned Real Property the respective certificates for land use rights and property ownership certificates, and has fully paid all the amounts due and payable for acquiring or constructing each Owned Real Property payable to any Person (including land granting fees,

various land acquisition fees, and project construction funds). With respect to each Owned Real Property: except as would not reasonably be expected to have a Material Adverse Effect, (i) none of the Group Company has leased or otherwise granted to any Person the right to use or occupy such Owned Real Property or any portion thereof and (ii) other than the statutory rights of first refusal of any lessee under the Laws of the PRC, there are no outstanding options, rights of first offer or rights of first refusal to purchase such Owned Real Property or any portion thereof or interest therein.

V

- (b) With respect to each of the Leases: except as would not reasonably be expected to have a Material Adverse Effect, (i) such Lease is legal, valid, binding, enforceable and in full force and effect, (ii) the Group Companies' possession of the Leased Real Property under such Lease has not been disturbed and, to the knowledge of the Company, there are no disputes with respect to such Lease, (iii) neither any Group Company nor, to the knowledge of the Company, any other party to the Lease is in material breach or default under such Lease, (iv) the Group Companies have a valid and subsisting leasehold interest in all property under the Leases, in each case free and clear of all Encumbrances except for Permitted Encumbrances, (v) none of the Group Companies has subleased, licensed or otherwise granted any Person the right to use or occupy such Leased Real Property or any portion thereof; and (vi) none of the Group Companies has collaterally assigned or granted any other security interest in such Lease or any interest therein.
- (d) Except as would not reasonably be expected to have a Material Adverse Effect, there is no material violation of any applicable Laws (including land-related Laws) relating to any Real Property of the Group Companies. The acquisition of, construction on and use of each Real Property (including the Real Property that is under construction) are in all material respects in compliance with all applicable Laws. There is no mortgage or other Encumbrance on any Real Property of the Group Companies that may have a Material Adverse Effect except for Permitted Encumbrances. There has been no material dispute, disagreement, claim or Action with respect to any Real Property of the Group Companies which may have a Material Adverse Effect. The Group Companies have not received any written notice from any relevant Governmental Authorities indicating that such Government Authorities have decided to expropriate, demolish, relocate or compulsorily purchase any Real Property of the Group Companies. The Group Companies may possess and use such Real Property in a safe and uninterrupted manner, and there are no contractual or legal restrictions that preclude or restrict the ability of the Group Companies to use such Real Property for the purposes for which it is currently being used, except in each case as would not materially and adversely impact the Group Companies, taken as a whole.
- (e) No party to any Lease has given written notice to any Group Company of, or made a written claim against any Group Company with respect to, any material breach or default thereunder. None of the Group Companies has received written notice of the existence of any outstanding Governmental Order, and, to the knowledge of the Company, there is no such Governmental Order threatened, relating to the ownership, lease, use, occupancy or operation by any Person of any Owned Real Property or Leased Real Property, which may have Material Adverse Effect.

# 11. Intellectual Property

(a) The Licensed Intellectual Property and the Owned Intellectual Property together constitute all the Intellectual Property used or held by the Group Companies for use in, the conduct of the business of the Group Companies. There exist no restrictions on the disclosure, use, license or transfer of the Owned Intellectual Property, that may have Material Adverse Effect. The consummation of the transactions contemplated by this Agreement will not alter, encumber, impair or extinguish any Owned Intellectual Property or Licensed Intellectual Property.

vi

- (b) To the knowledge of the Company, none of the Group Companies has infringed, misappropriated or otherwise violated any Intellectual Property of any third party, which may have Material Adverse Effect. There is no Action pending against or, to the knowledge of the Company, threatened against or affecting any of the Group Companies, that may have Material Adverse Effect.
- (d) The Group Companies are the owners of all Owned Intellectual Property and hold relevant right, title and interest in and to all Owned Intellectual Property and the Licensed Intellectual Property, free and clear of any Encumbrances except for Permitted Encumbrances. None of the Owned Intellectual Property and Licensed Intellectual Property material to the operation of the business of the Group

Companies has been adjudged invalid or unenforceable in whole or part, and, to the knowledge of the Company, all such Owned Intellectual Property and Licensed Intellectual Property are valid and enforceable.

- (e) The Group Companies have taken all actions necessary to maintain and protect the Owned Intellectual Property and their rights in Licensed Intellectual Property.
- (f) To the knowledge of the Company, no Person has infringed, misappropriated or otherwise violated any Owned Intellectual Property or Licensed Intellectual Property. The Group Companies have taken reasonable steps in accordance with normal industry practice to maintain the confidentiality of all Intellectual Property that is material to the business or operation of the Group Companies and the value of which to any of the Group Companies is contingent upon maintaining the confidentiality thereof.
- (g) To the extent that any Intellectual Property has been developed or created by a third party (including any current or former employee of any of the Group Companies) for the business of the Group Companies, the Group Companies thereby either (i) have obtained ownership of and is the exclusive owner of, or (ii) have obtained a valid and unrestricted right to exploit, sufficient for the conduct of its business, such Intellectual Property.
- (h) The Information Technology is functional in all material respects and operate and perform in a manner that permits the Group Companies to conduct their respective businesses. The Group Companies have taken commercially reasonable and necessary actions to protect the confidentiality, integrity, operation and security of the Information Technology (and all information and transactions stored or contained therein or transmitted thereby) against any unauthorized, use, access, interruption, malfunction, modification, or corruption. There has been no material unauthorized use, access, interruption, modification, corruption or malfunction of any information technology (or any information or transactions stored or contained therein or transmitted thereby).
- (k) The Group Companies have at all times complied with all applicable Laws relating to privacy, data protection and the collection and use of personal information and user information gathered or accessed in the course of the operation of the Group Companies in all material aspects. The Group Companies have at all times complied in all material aspects with all rules, policies and procedures established by the Group Companies from time to time with respect to the foregoing. To the knowledge of the Company, no claims have been asserted or threatened against, or would likely to be asserted or threatened against any of the Group Companies by any Person alleging a violation of such Person's privacy, personal or confidentiality rights under any such laws, regulations, rules, policies or procedures. Subject to the satisfaction of relevant condition precedent, the consummation of the transactions contemplated by this Agreement will not breach or otherwise cause any violation of any such laws, regulations, rules, policies or procedures, except where such breach would not reasonably be expected to cause a Material Adverse Effect.

vii

- 12. Material Contracts
- (a) The Company has made the Buyer available to the Material Contracts.
- (b) "Material Contracts" means any of the following:
  - (i) any Contract relating to the issuance of any equity securities of any Group Company;
  - (ii) any Contract with an aggregate contract value or involving payments (or a series of payments) to or from any Person, of RMB50 million or more;
  - (iii) any partnership, joint venture, strategic cooperation or other similar Contract;
  - (iv) any Contract involving a loan (other than accounts receivable in the ordinary course of business) or advance to, or investment in, any Person in the amount in excess of RMB50 million;
  - (v) any Contract relating to the acquisition or disposal of any business (whether by merger, sale of stock, sale of assets or otherwise) with a fair market value or purchase price in excess of RMB50 million;

- (vi) any Contract relating to indebtedness of any Group Company or any guarantee of indebtedness by any Group Company (in either case, whether incurred, assumed, guaranteed or secured by any asset) in excess of RMB50 million;
- (vii) any Contract involving the waiver, compromise or settlement of any material Action;
- (viii) any Contract that limits the freedom of any Group Company to compete in any material line of business or with any Person or in any geographic area or which would so limit the freedom of any Group Company after the Closing Date;
- any Contract, in excess of RMB 50 million, between any Group Company on the one hand and (A) any Affiliate of any Group Company (other than the other Group Companies), (B) any director or officer of any Group Company or of any Person described in clause (A), or (C) any Affiliate of any natural person described in clause (A) or (B), on the other hand, except for the employment agreements relating to services as employees, officers or directors of any Group Company;
- (x) any Contract with Governmental Authorities material to the operation of any Group Company party thereto in excess of RMB50 million;

viii

- (xi) any Contract that involves prohibition of payment of dividends or distributions in respect of the equity interests of any Group Company;
- any Contract with an amount in excess of RMB50 million that will be terminated or varied upon a change of control involving any Group Company or the consummation of the transactions contemplated by this Agreement, will subject such change of control or the transactions contemplated by this Agreement to the consent of any Person, or will trigger any payment to any Person as a result of such change of control or the consummation of the transactions contemplated by this Agreement;
- (xiii) any other Contract that is material to the Group Companies.
- (c) Each Material Contract (i) is legal, valid and binding on each party thereto, and is in full force and effect and enforceable against each party thereto in accordance with its terms; and (ii) will remain in full force and effect and will not result in any penalty or have other material adverse effect upon the completion of the transactions contemplated by this Agreement. None of the Group Companies is in any material respect in breach of or default under any Material Contract, and to the knowledge of the Company, nor is any other party thereto in any material respect in breach thereof or in default thereunder. None of the Group Companies has received any notification with respect to termination, revocation of or default under any Material Contract. To the knowledge of the Company, no Person intends to terminate or cancel any Material Contract, and no fact or event exists that would give rise to any claim of default which may have Material Adverse Effect under any Material Contract.

#### 13. Related Party Transactions

Each existing transaction between, on one hand, a Group Company, and, on the other hand, another Group Company or the shareholders, officers, directors or employees of any Group Company, or the Affiliates of any of them (including transfer of tangible assets, transfer of financial assets, transfer of the right to use or the ownership of intangible assets, financial transactions, services transactions etc.) (i) has the necessity and true commercial intent, (ii) does not prejudice the legal interests of the Group Companies, (iii) is undertaken at a price based on arm's length principle, and (iv) has been properly disclosed as required pursuant to applicable Laws.

- 14. Compliance with Laws; Permits
- (a) Within the 3-year period prior to the Effective Date, each Group Company is in all material respects in compliance with all applicable Laws (including without limitation Anticorruption Law and Laws with respect to AMR registration, hygiene, fire prevention, development, construction and use of land, and environment protection, taxation or labour) and Governmental Order. In the past three (3) years, none of the Group Companies has in any material respects violated any of applicable Laws or Governmental Orders.
- (b) Each Group Company has been granted and owns all franchises, permits, licenses, certificates, consents, approvals and any similar authorizations of any Governmental Authority which is necessary for the conduct of their business (the "Material Company Permits").

None of the Group Companies is in any material respects in default under any Material Company Permit. All of the Material Company
Permits are in full force and effect, and no suspension or cancellation of any of the Material Company Permits is pending.

ix

- (c) In the past three (3) years, no Group Company, or to the knowledge of the Company, any of their respective directors, officers, employees or other persons with formal authorization of the Group Company and acting on behalf of any Group Company, has violated any Anticorruption Law in any material respects.
- (d) In the past three (3) years, no Group Company has conducted or initiated any internal investigation or made a voluntary, directed, involuntary or other disclosure to any Governmental Authority with respect to any alleged act or omission arising under or relating to any noncompliance with any Anticorruption Law, nor to the knowledge of the Company, any of their respective directors, officers, employees or other persons with formal authorization of the Group Company and acting on behalf of any Group Company has received any written notice, request, allegation or citation from a Governmental Authority for any noncompliance with any Anticorruption Law. The Group Companies have implemented necessary policies and procedures designed to ensure that each Group Company and the employees and other intermediaries of the Group Companies comply in all material respects with applicable Anticorruption Laws. The Group Companies have implemented and maintain effective internal controls necessary and reasonably designed to prevent and detect violations of Anticorruption Laws.
- (e) No Group Company, or to the knowledge of the Company, any of their respective directors, officers, employees or other persons with formal authorization of the Group Company and acting on behalf of any Group Company is a Prohibited Person. No Group Company has engaged in any business or dealings involving or relating to (i) a Sanctioned Jurisdiction; or (ii) a Prohibited Person, in each case, to the extent such activities would violate applicable Laws.
- (f) The operations of the Group Companies are and have been conducted at all times in compliance with all Anti-Money Laundering Laws. The Group Companies have instituted, maintain and enforce necessary policies and procedures designed to ensure compliance with Anti-Money Laundering Laws. For the past three (3) years, none of the Group Companies has been penalized for or threatened to be charged with, or given notice of any violation of, or, to the knowledge of the Company, under investigation with respect to, any Anti-Money Laundering Laws, and no Action by or before any court, Governmental Authority or any arbitrator involving any of the Group Companies with respect to Anti-Money Laundering Laws is pending or threatened.
- 15. Labor and Employment Matters
- (a) There is no ongoing strikes or collective labor disputes between any Group Company and its employees that may have Material Adverse Effect.
- (b) Except for what has been disclosed, each of the Group Companies has complied in all material respects with all applicable Laws relating to employment or labor relations. There is no material dispute pending and that may have Material Adverse Effect between any Group Company and any of its existing or former employees, directors, officers or trade union, representative council or other body representing any of them.
- (c) Each Key Person has entered into an employment agreement and a non-compete and non-solicitation agreement or an employment agreement containing non-compete and non-solicitation provisions, and provisions for the protection of the Group Companies' confidential information, in each case with the relevant Group Company.

X

(d) To the knowledge of the Company, no important officer or key employee (including the Key Persons) intends to terminate their employment with any Group Company, that may have Material Adverse Effect. To the knowledge of the Company, no Group Company has a present intention to terminate the employment of any important officer or key employee (including the Key Persons), that may have a Material Adverse Effect.

(e) The performance of each Key Person's duties to the Group Companies will not, to the knowledge of the Company, constitute a breach of, or otherwise contravene, the terms of any employment or other agreement or policy, or any judgment, decree or order of any court or administrative agency, to which he/she is a party or is otherwise bound.

## 16. Environmental Matters

- (a) Each of the Group Companies has been in material respects in compliance with applicable Environmental Laws.
- (b) None of the Group Companies has received any notice or Governmental Order from any Governmental Authority, relating to (i) any actual or alleged violation of or failure to comply with any Environmental Laws in material respects, or (ii) any compulsory request for information pursuant to any Environmental Laws or any actual and material Liability resulting from or arising under any Environmental Laws, or (iii) a requirement to investigate or remediate Hazardous Materials.
- (b) There are no pending or, to the knowledge of the Company, threatened Actions resulting from or arising under or pursuant to applicable Environmental Laws with respect to or affecting any of material facilities of the Group Companies or, against any Person whose Liability for such Action the Group Companies have retained or assumed contractually and there are no facts, conditions, situations or set of circumstances which could reasonably be expected to result in or be the basis for any such liability.
- (d) With respect to the Group Companies, except as would not reasonably be expected to have a Material Adverse Effect, there is no Hazardous Materials present on or under the facilities in violation of applicable Environmental Laws. With respect to the Group Companies, except as would not reasonably be expected to have a Material Adverse Effect, no Hazardous Materials have been generated, treated, stored, discharged, released, spilled or disposed at, on, under or from the facilities in violation of, or in a manner or in a location that could result in material Liability under any applicable Environmental Laws.

#### 17. Grants, Incentives and Subsidies

The Group Companies are in all material respects in compliance with the terms and conditions of all outstanding grants, incentives, exemptions and subsidies from the Governmental Authorities or any agency thereof, or from any foreign governmental or administrative agency, granted to any Group Company ("Grants"), and has duly fulfilled all material and compulsory undertakings required thereby to be fulfilled prior to the date hereof. The Grants are provided in full compliance with applicable Laws. To the knowledge of the Company, there is no event or other set of circumstances which would reasonably be expected to lead to the revocation, material modification or claw-back of any of the Grants that have been approved or received, and such revocation, material modification or claw-back may have Material Adverse Effect.

хi

#### 18. Insurance

The Group Companies have taken out and maintained in full force and effect adequate and necessary insurance policies or binding slips (other than those expired pursuant to the terms thereof in the ordinary course) with a reputable and reliable insurer or insurers, against such risks and in such amounts and having such coverage as common practice in the automobile industry. All premiums due with respect to such insurance policies have been paid in accordance with the terms thereof.

# 19. Litigation

There is no Action pending or, to the knowledge of the Company, alleged or threatened, against any Group Company or any share, security, equity interest, material property or asset of any Group Company, or to which the a Group Company is otherwise a party, before any Governmental Authority; nor to the knowledge of the Company, is there any reasonable basis for any such Action, except for any such Action that would not materially and adversely impact the Group Companies, taken as a whole. None of the Group Companies, is a party to or subject to the provisions of any order, writ, injunction, judgment or decree of any court or government agency or instrumentality that may have Material Adverse Effect, and there is no Action by any Group Company currently pending and which may have Material Adverse Effect.

#### 20. Information

	xii		

# **Subscription Agreement**

for

# LYNK & CO AUTOMOTIVE TECHNOLOGY CO., LTD.

by and among

# ZHEJIANG ZEEKR INTELLIGENT TECHNOLOGY CO., LTD.

NINGBO GEELY AUTOMOBILE INDUSTRY CO., LTD.

and

# LYNK & CO AUTOMOTIVE TECHNOLOGY CO., LTD.

**Dated Nov 14, 2024** 

THIS SUBSCRIPTION AGREEMENT (this "Agreement") is entered into on November 14, 2024 (the "Effective Date") by and among:

- ZHEJIANG ZEEKR INTELLIGENT TECHNOLOGY CO., LTD. (浙江极氪智能科技有限公司), a limited liability company established in the PRC with its address at Room 1031, Shangwu Building 1, No. 1388 Minshan Road, Beilun District, Ningbo, Zhejiang Province, the PRC ("Zhejiang ZEEKR" or "Subscriber");
- NINGBO GEELY AUTOMOBILE INDUSTRY CO., LTD. (宁波吉利汽车实业有限公司), a limited liability company established in the PRC with its address at No. 918 Binhai Fourth Road, Hangzhou Bay New Area, Ningbo, Zhejiang Province, the PRC ("Ningbo Geely"); and
- LYNK & CO AUTOMOTIVE TECHNOLOGY CO., LTD. (领克汽车科技有限公司), a limited liability company established in the PRC with its address at 201-206, No. 918 Binhai Fourth Road, Andong Town, Qianwan New District, Ningbo, Zhejiang Province, the PRC (the "Company").

Each of the parties listed above is referred to hereinafter individually as a "Party" and collectively as the "Parties".

# RECITALS

**WHEREAS**, the Company was incorporated on October 20, 2017. The Company and its Subsidiaries are principally engaged in the development, manufacturing and sale of LYNK & CO brand vehicles and/or related parts. As of the date of this Agreement, the registered capital of the Company is RMB 7,500,000,000.

WHEREAS, the Company wishes to increase its registered capital from RMB 7,500,000,000 to RMB 7,653,061,225 with the newly increased registered capital of RMB 153,061,225 ("Increased Registered Capital"). The Company hereby agrees to issue and sell to Zhejiang ZEEKR, and Zhejiang ZEEKR hereby agrees to subscribe for and purchase from the Company the Increased Registered Capital upon the terms and conditions of this Agreement ("Subscription").

WHEREAS, concurrently with the execution of this Agreement, Zhejiang ZEEKR, Zhejiang Geely Holdings Group Co., Ltd. (浙江 吉利控股集团有限公司) ("GH"), Volvo Cars (China) Investment Co., Ltd. (沃尔沃汽车(中国)投资有限公司) ("VCI") and the Company entered into the Equity Transfer Agreement ("ETA"), pursuant to which GH agrees to transfer 20% of equity interests in the Company (corresponding to registered capital RMB 1,500,000,000 of the Company) to Zhejiang ZEEKR and VCI agrees to transfer 30% of equity interests in the Company (corresponding to registered capital RMB 2,250,000,000 of the Company) to Zhejiang ZEEKR ("Equity Transfer").

WHEREAS, after consummation of the Equity Transfer and the Subscription, Zhejiang ZEEKR will own 51% of the equity interests in the Company.

1

**NOW, THEREFORE**, in consideration of the representations, warranties, covenants and agreements hereinafter contained, the Parties agree as follows:

# 1. <u>DEFINITIONS AND INTERPRETATION</u>

- 1.1 Capitalized terms used in this Agreement shall have the following meanings.
- "Action" means any litigation, hearing, claim, suit, charge, action, complaint, arbitration, settlement, decision, inquiry, investigation or other proceedings initiated by or before any Governmental Authority.
- "Affiliate" of a Person means any other Person that, directly or indirectly, Controls, is Controlled by, or is under common Control with such Person; provided that, for the purpose of this Agreement, Ningbo Geely shall not be deemed as an Affiliate of the Company and vice versa.
  - "Agreement" has the meaning ascribed thereto in the Preamble.
  - "Amended Articles" has the meaning ascribed thereto in Section 3.2.1(e).
  - "AMR" means the State Administration for Market Regulation of the PRC or its competent local counterparts.
  - "AMR Registration" has the meaning ascribed thereto in Section 4.1.4.
- "Business Day(s)" means any day that is not a Saturday, a Sunday or other day on which commercial banks are required or authorized by applicable Law to be closed in the PRC.
  - "CIETAC" has the meaning ascribed thereto in Section 11.2.2.
  - "Closing" has the meaning ascribed thereto in Section 3.1.
  - "Closing Date" has the meaning ascribed thereto in Section 3.1.
  - "Company" means the meaning ascribed thereto in the Preamble.
  - "Confidential Information" has the meaning ascribed thereto in Section 9.1.
- "Control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management of a Person, whether through the ownership of voting securities, by contract, credit arrangement or proxy, as trustee, executor or agent or otherwise. For purposes of this definition, a Person shall be deemed to Control another Person if such first Person, directly or indirectly, owns or holds more than 50% of the voting equity securities in such other Person for the general election of directors, or if such first Person, directly or indirectly, controls the board of directors, managing partner or other similar governing body or position of such other Person. The terms "Controlled" and "Controls" shall have meanings correlative to the foregoing.
  - "Dispute" has the meaning ascribed thereto in Section 11.2.1.

"Encumbrance" means any lien, pledge, hypothecation, mortgage, security interest, charge, claim, title retention, levy, proxy, option, restrictive covenant, license, right of first refusal, right of first offer, easement, or other encumbrance or adverse claim of any kind (including without limitation any conditional sale agreement, capital lease or other title retention agreement relating to property or asset).

"Equity Transfer" has the meaning ascribed in the Recitals.

"<u>ETA</u>" means the Equity Transfer Agreement in relation to the transfer of equity interests in the Company executed by and among the Company, Zhejiang ZEEKR, GH and VCI on the Effective Date.

"GA" means Geely Automobile Holdings Limited (吉利汽車控股有限公司), a company established in the Cayman Islands with its shares listed on the Main Board of the Stock Exchange of Hong Kong Limited.

"GH" has the meaning ascribed thereto in the Recitals.

"Governmental Authority" means any nation or government or any province or state or local or any other political subdivision thereof, or any entity, authority or body exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government, including any government authority, agency, department, board, commission or instrumentality or any political subdivision thereof, any court or arbitrator, any arbitration tribunal, and any self-regulatory organization or national or international stock exchange.

"Governmental Order" means any applicable order, ruling, decision, verdict, decree, writ, subpoena, mandate, precept, command, directive, consent, approval, award, judgment, injunction or other similar determination or finding by, before or under the supervision of any Governmental Authority.

"Group Companies" means collectively, the Company and its Subsidiaries currently and hereafter, and each is herein referred to individually as a "Group Company".

"Increased Registered Capital" has the meaning ascribed in the Recitals.

"Intellectual Property" means any and all (a) patents (including all reissues, divisionals, provisionals, continuations, continuations in part, re-examinations, renewals and extensions thereof), patent applications, and other patent rights, (b) trademarks, service marks, tradenames, brand names, logos, slogans, trade dress, design rights, and other similar designations of source or origin, together with all goodwill associated with any of the foregoing and applications, registrations and renewals in connection therewith, (c) copyrights, mask works, and copyrightable works, and all applications, registrations for and renewals in connection therewith, (d) internet domain names, web addresses, web pages, websites and related content, accounts with social media companies and the content found thereon and related thereto, and uniform resource locators, (e) proprietary Software, including source code, object code and supporting documentation for such Software, (f) trade secrets and proprietary information, including confidential business information, technical data, customer lists, data collections, methods and inventions (whether or not patentable and where or not reduced to practice), (g) copies and tangible embodiments of any of the foregoing, (h) all other intellectual property, whether or not registrable, in each case, under any Law or statutory provision or common law doctrine in any country, and (i) all rights to sue or recover and retain damages and costs and attorneys' fees for past, present and future infringement or misappropriation of any of the foregoing.

3

"Interim Period" has the meaning ascribed thereto in Section 6.1.

"Joint Venture Contract" has the meaning ascribed thereto in Section 4.1.11.

"<u>Law(s)</u>" means any federal, state, territorial, foreign or local law, common law, statute, ordinance, rule, regulation, code, constitution, treaty, measure, notice, circular, judgment, decree, opinion, Governmental Order or other requirement or rule of law of any Governmental Authority, including any rules promulgated by a stock exchange or regulatory body.

"Long Stop Date" has the meaning ascribed thereto in Section 4.3.

"Loss(es)" means, without duplication, any and all demands, losses, damages, penalties, judgments, fines, taxes, claims, liabilities and obligations of any kind or nature whatsoever, including costs and expenses of investigating, proceedings, preparing or defending any such claim or Action and legal fees and expenses in connection therewith, and any amount paid in settlement of, any pending or threatened Action, but in each case excluding indirect, incidental, special, exemplary, punitive, consequential loss, any loss of profits, loss of production, loss of revenue, business interruption, loss of data or loss of business information arising out of, based on or resulting from this Agreement.

"Material Adverse Effect" means any event, fact, circumstance, occurrence, development, change or effect that, individually or in the aggregate with all other events, facts, circumstances, occurrences, developments, changes or effects, has, or would reasonably be expected to have a material adverse effect on (a) the business, conditions, assets, liabilities (including but not limited to contingent liabilities), operations or financial conditions of the Group Companies, taken as a whole; or (b) the qualifications or abilities of the Group Companies, taken as a whole, in carrying out the business currently engaged by the Group Companies; or (c) the ability of the Company to consummate the transactions contemplated by this Agreement; provided, however, that in determining whether a Material Adverse Effect has occurred pursuant to sub-clauses (a) or (b) above, there shall be excluded any effect on the Group Companies to the extent arising out of any of the following: (i) any action required to be taken pursuant to the terms and conditions of this Agreement or with the written consent of the Subscriber; (ii) the announcement of the transactions contemplated by this Agreement; (iii) any changes in applicable Laws or the interpretation or enforcement thereof or in applicable accounting principles or the interpretation thereof; or (iv) any pandemic, earthquake, typhoon, other natural disasters, any outbreak or escalation of hostilities of war or any act of terrorism; except in the case of sub-clauses (iii) or (iv), to the extent that the Group Companies, taken as a whole, are disproportionately affected thereby as compared with other participants in the same industries and geographic markets in which the Group Companies operate.

"Ningbo Geely" has the meaning ascribed thereto in the Recitals.

"Permitted Encumbrances" means (i) statutory liens for current taxes, special assessments or other governmental or quasi-governmental charges not yet delinquent or the amount or validity of which is being contested in good faith by appropriate proceedings, (ii) mechanics', materialmen's, carriers', workers', warehousemen's, repairers' and similar statutory liens arising or incurred in the ordinary course of business, (iii) zoning, entitlement, building and other land use regulations imposed by any Governmental Authority, (iv) covenants, conditions, restrictions, easements and other similar matters of record affecting title to any real property which do not materially impair the occupancy or use of such real property for the purposes for which it is currently used or proposed to be used in connection with the Group Companies' businesses, (v) liens incurred or deposits or pledges made in connection with, or to secure payment of, worker's compensation, unemployment insurance, old age pension programs, social security, retirement and similar laws, and (vi) non-exclusive licenses of Intellectual Property entered into in the ordinary course of business.

4

"Person" means an individual, corporation, limited liability company, partnership, company, joint venture, association, trust, unincorporated organization, Governmental Authority, or other entity.

"PRC" means the People's Republic of China, for purpose of geographical and jurisdictional reference in this Agreement, excluding the Hong Kong Special Administrative Region, the Macau Special Administrative Region and Taiwan.

"Representatives" has the meaning ascribed thereto in Section 9.1.

"RMB" means Renminbi, the lawful currency of the PRC.

"Subscription" has the meaning ascribed thereto in the Recitals.

"Subscription Price" has the meaning ascribed thereto in Section 2.2.

"<u>Subsidiary</u>" means, with respect to any Person, any corporation, partnership, limited liability company or other organization, whether incorporated or unincorporated, which is Controlled by such Person.

"Target Equity" has the meaning ascribed in the ETA.

"VCI" has the meaning ascribed thereto in the Recitals.

- 1.2 Headings are inserted for convenience only and shall not affect the construction of this Agreement.
- 1.3 Unless the express context otherwise requires, references to:
  - Sections and Schedules are sections in and schedules to this Agreement and the Recitals and Schedules to this Agreement shall be deemed to form part of this Agreement;
  - 1.3.2 writing shall include any methods of producing or reproducing words in a legible and non-transitory form;
  - 1.3.3 the singular number shall include the plural and vice versa;
  - 1.3.4 "days" shall mean calendar days unless Business Days are expressly specified;

5

- 1.3.5 any Person shall include such Person's successors and permitted assigns; and
- any agreement or instrument shall be construed as a reference to that agreement or instrument as amended, novated or supplemented.
- The words "hereof," "hereby," "hereto," "herein," and "hereunder" and words of similar import, when used in this Agreement, shall refer to this Agreement as a whole and not to any particular provision of this Agreement.
- 1.5 Wherever the word "include," "includes" or "including" is used in this Agreement, it shall be deemed to be followed by the words "without limitation".
- 1.6 The word "or" shall not be exclusive.
- The word "extent" and the phrase "to the extent" shall mean the degree to which a subject or other thing extends, and such word or phrase shall not simply mean "if".
- In construing this Agreement, general words shall not be given a restrictive meaning by reason of the fact that they are followed by examples intended to be embraced by the general words.
- References to statutory provisions, including any subordinate legislation made under the relevant statute, shall be construed as references to those provisions as amended or re-enacted or as their application is modified by other provisions (whether before or after the date hereof) from time to time and shall include any provisions of which they are re-enactment (whether with or without modification).
- When calculating the period of time before which, within which or following which any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded and if the last day of such period is not a Business Day, the period shall end on the next succeeding Business Day.
- The Parties have each participated in the negotiation and drafting of this Agreement and if an ambiguity or question of interpretation should arise, this Agreement shall be construed as if drafted jointly by the Parties and no presumption of burden of proof shall arise favoring or burdening any Party by virtue of the authorship of any provision in this Agreement.

# 2. <u>SUBSCRIPTION</u>

- Subject to the fulfilment and/or waiver of the conditions precedent set out in Section 4.1 and 4.2, the Subscriber agrees to subscribe for and purchase from the Company, and the Company agrees to allot and issue to the Subscriber, the Increased Registered Capital.
- Subject to the fulfilment and/or waiver of the conditions precedent set out in Section 4.1 and 4.2, the price to be paid by the Subscriber to the Company for the Increased Registered Capital shall be RMB 367,346,940 (the "Subscription Price"), which shall be paid in cash on the Closing Date. The portion of the Subscription Price in excess of the Increased Registered Capital shall be credited to the capital reserve of the Company.

6

Immediately after the Closing of the Subscription and the closing of the Equity Transfer contemplated in the ETA, the registered capital of the Company shall be increased from RMB 7,500,000,000 to RMB 7,653,061,225 and the shareholding structure of the Company shall be as follows:

Shareholder	Registered Capital Subscribed by such Shareholder	Shareholding Proportion
Zhejiang ZEEKR	RMB 3,903,061,225	51%
Ningbo Geely	RMB 3,750,000,000	49%
In Total	RMB 7,653,061,225	100%

2.4 Upon the Closing, the Subscriber shall be entitled to all rights and interests attaching to the Increased Registered Capital as the shareholder of the Company in accordance with PRC Laws, Joint Venture Contract and the Amended Articles.

## 3. <u>CLOSING</u>

The closing of the Subscription (the "Closing") shall take place remotely via the electronic exchange of the closing documents and signatures on a date mutually agreed on by the Parties but in any event no later than fifteenth (15<sup>th</sup>) Business Days following the date on which all conditions precedent set forth in Sections 4.1 and 4.2 below are satisfied or waived in writing by the relevant Party (other than conditions which by their nature are to be satisfied at the Closing), or at such other time and date as the Parties may otherwise mutually agree in writing (the date of the Closing, the "Closing Date").

The Parties agree that, the Closing shall consummate immediately after the closing of the Equity Transfer in accordance with the terms of the ETA.

- 3.2 Closing Payment and Deliverables
  - 3.2.1 At or before the Closing Date, the Company and Ningbo Geely shall, and Ningbo Geely shall cause the Company to deliver to the Subscriber:
    - (a) a closing certificate executed by the Company and Ningbo Geely, dated the Closing Date, certifying that each of the conditions specified in Section 4.1 have been satisfied or waived;
    - (b) documentary evidence of the completion of the AMR Registration, including the updated business license of the Company and any other notification document issued by the competent AMR in relation to the Subscription;
    - (c) the capital contribution certificate duly executed and issued by the Company to the Subscriber reflecting that the Increased Registered Capital having been registered under the name of the Subscriber;
    - (d) a copy of the updated register of shareholders duly executed by the Company, which reflects updated shareholding structure of the Company upon the consummation of the Subscription and the Equity Transfer;

- copies of the resolutions of board of directors and shareholders meeting of the Company approving (i) the Subscription, (ii) adoption of amendment to the article of association of the Company which, among other things,
- (e) reflects the completion of the Equity Transfer and the Subscription, in the form attached as Exhibit A hereto (the "Amended Articles"), and (iii) the change of legal representative, director(s), supervisor(s), and/ or general manager of the Company hereto and related matters; and
- (f) a copy of the waiver of pre-emptive right by relevant shareholders of the Company to waive their pre-emptive right with respect to the Subscription.
- 3.2.2 At the Closing Date, the Subscriber shall pay and deliver to the Company:
  - (a) the Subscription Price (which may be evidenced by a copy of the receipt or payment confirmation from the bank) to the bank account of the Company; and
  - (b) a closing certificate executed by the Subscriber, dated the Closing Date, certifying that each of the conditions specified in Section 4.2 has been satisfied.

# 4. <u>CONDITIONS PRECEDENT</u>

4.1 Conditions Precedent to Obligations of the Subscriber

The obligation of the Subscriber to consummate the Subscription contemplated by this Agreement is subject to the fulfillment, on or prior to the Closing Date, of each of the following conditions precedent (any or all of which may be waived by the Subscriber in whole or in part to the extent permitted by applicable Laws):

- GA's Approval. The transactions contemplated under this Agreement, and the Equity Transfer shall have been duly approved by the independent shareholders of GA, in compliance with applicable Laws and the Rules Governing the Listing of Securities on The Stock Exchange of Hong Kong Limited.
- Company's Internal Approval. The Company shall have duly approved (i) implementation of the transactions contemplated by this Agreement and the Equity Transfer, and (ii) adoption of Amended Articles and (iii) the change of
- 4.1.2 legal representative, director(s), supervisor(s), and/or general manager of the Company, in accordance with applicable Laws and its joint venture contract and articles of association then in effective, and reasonable evidence shall have been delivered to the Subscriber.
- Waiver of Pre-emptive Right. The existing shareholder(s) of the Company shall have waived their pre-emptive right with respect to the Subscription and right of first refusal with respect to the Equity Transfer, and reasonable evidence shall have been delivered to the Subscriber.
- 4.1.4 AMR Registration. The Company shall have completed the filing and registration procedures with competent AMR in connection with the Subscription and Equity Transfer, including but not limited to (i) Target Equity in the ETA and the Increased Registered Capital having been registered under the name of the Subscriber; and (ii) Amended Articles having been filed (the "AMR Registration"), and reasonable evidence shall have been delivered to the Subscriber.

8

- 4.1.5 <u>Completion of the Equity Transfer.</u> The closing of the Equity Transfer contemplated in the ETA shall have consummated in accordance with the terms of the ETA.
- 4.1.6 Accuracy of Representations and Warranties. The representations and warranties made by the Company and Ningbo Geely in this Agreement shall be true and correct as of the date of this Agreement and as of the Closing Date as though

made on and as of the Closing Date except for representations and warranties made only at and as of a certain date, in which case as of such specified date.

- Performance of Agreement. The Company and Ningbo Geely shall have performed and complied in all material respects with all obligations and agreements required in this Agreement to be performed or complied with by them on or prior to the Closing Date.
- 4.1.8 No Material Adverse Effect. No Material Adverse Effect shall have occurred since the Effective Date.
- Third-Party Consent. The Company and Ningbo Geely shall have obtained all other consents, approvals, orders or authorization of, or registration, declaration or filing with, any Governmental Authority or third party required to be obtained or made by it in connection with the consummation of the transactions contemplated hereby.
- No Injunction. There shall be no injunction, restraining order or decree of any nature of any Governmental Authority or applicable Laws that is in effect that restrains or prohibits the consummation of the transactions contemplated by this Agreement.
- 4.1.11 <u>Joint Venture Contract</u>. A copy of the shareholders agreement ("**Joint Venture Contract**") in the form attached as Exhibit B hereto shall have been delivered to the Subscriber, duly executed by the Subscriber and Ningbo Geely.
- 4.1.12 Closing Certificate. The Company and Ningbo Geely shall have delivered to the Subscriber a copy of the closing certificate set forth in Section 3.2.1(a) in form and substance reasonably satisfactory to the Subscriber.

C

# 4.2 Conditions Precedent to Obligations of the Company

The obligation of the Company to consummate the Subscription contemplated in this Agreement is subject to the fulfillment, on or prior to the Closing Date, of each of the following conditions precedent (any or all of which may be waived by the Subscriber in whole or in part to the extent permitted by applicable Laws):

- Accuracy of Representations and Warranties. The representations and warranties made by the Subscriber in this Agreement shall be true and correct as of the date of this Agreement and as of the Closing Date as though made on and as of the Closing Date, except for representations and warranties made only at and as of a certain date, in which case as of such specified date.
- Performance of Agreement. The Subscriber shall have performed and complied with, in all material respects, all obligations and agreements required in this Agreement to be performed or complied with by it on or prior to the Closing Date.
- No Injunction. There shall be no injunction, restraining order or decree of any nature of any Governmental Authority or applicable Laws that is in effect that restrains or prohibits the consummation of the transactions contemplated by this Agreement.
- Closing Certificate. The Subscriber shall have delivered to the Company and Ningbo Geely a copy of the closing certificate set forth in Section 3.2.2(b) in form and substance reasonably satisfactory to the Company and Ningbo Geely.

## 4.3 Long Stop Date

The Parties jointly undertake to use their reasonable best efforts to procure the fulfilment of all the conditions precedents and the occurrence of the Closing as soon as possible but in any event within twelve(12) months after the Effective Date (the "Long Stop Date") or such later date as may be agreed by the Parties in writing.

# 5. <u>REPRESENTATIONS AND WARRANTIES</u>

- The Company and Ningbo Geely, jointly and severally, hereby represent and warrant to the Subscriber that as of the date hereof and as of the Closing Date and each of the dates between the date hereof and the Closing Date (except if a representation or warranty is made as of a specified date, as of such date),
  - (a) such Party is a corporation duly and legally organized, validly existing and in good standing under applicable Laws;
  - such Party has all requisite power, authority and approvals required to enter into this Agreement and has all requisite power, (b) authority and approvals to perform fully each and every one of its obligations thereunder in accordance with the applicable Laws;
    - such Party's execution and performance of this Agreement do not and will not, in any material respects, (i) violate, conflict with or result in the breach of any provision of its articles of association or similar charter documents, (ii) conflict with or violate any applicable Laws or Governmental Order applicable to it, or (iii) conflict with, result in any breach of, constitute a default (or an event which with the giving of notice or lapse of time, or both, would become a default) under,
  - (c) require any consent under, or give to others any rights of termination, amendment, acceleration, suspension, revocation or cancellation of any contract to which such Party is a party or result in the creation of any Encumbrance (except for Permitted Encumbrance) upon any of the properties or assets of such Party, except where in each case the failure to obtain such consent, approval, authorization, action or to make such filing or notification would not, individually or in the aggregate, prevent or materially delay the performance of such Party's obligations under this Agreement;

10

- such Party's execution and performance of this Agreement do not and will not require any consent, approval, authorization or other order of, action by, filing with, or notification to, any Governmental Authority, except (i) the registration and filing (d) procedure by the Company with Ningbo Administration for Market Regulation, and (ii) where in each case the failure to obtain such consent, approval, authorization, action or to make such filing or notification would not, individually or in the aggregate, prevent or materially delay the performance of such Party's obligations under this Agreement;
- such Party's representative whose signature is affixed to this Agreement is fully authorized to sign this Agreement on behalf of such Party, and this Agreement has been duly executed and delivered by such Party and assuming the due authorization, execution and delivery by the other Parties, constitutes the legal, valid, binding and enforceable obligations of such Party in accordance with the terms hereunder; and
- the Increased Registered Capital, when issued in accordance with the terms and conditions of this Agreement and sold against receipt of consideration therefor, will be validly issued and free from all charges, liens, Encumbrance or other third party rights, claims or interests, except for the pre-emptive right under the Company Law of the PRC. Other than the Subscription and the Equity Transfer, there are no preemptive rights, options, call warrants, convertible securities or other contracts, arrangements or undertakings under which the Company is or may become obligated to issue, assign or transfer any equity securities of the Company.
- The Subscriber hereby represents and warrants to the Company and Ningbo Geely that as of the date hereof and as of the Closing

  Date and each of the dates between the date hereof and the Closing Date (except if a representation or warranty is made as of a specified date, as of such date),
  - (a) it is a corporation duly and legally organized, validly existing and in good standing under the PRC Laws;
  - except as otherwise described in this Agreement, it has all requisite power, authority and approvals required to enter into (b) this Agreement and has all requisite power, authority and approvals to perform fully each and every one of its obligations thereunder in accordance with the applicable Laws;
  - its execution and performance of this Agreement does not and will not (i) violate, conflict with or result in the breach of any provision of its articles of association or similar charter documents, (ii) conflict with or violate any applicable Laws or Governmental Order applicable to it, in each case in any material respect, (iii) conflict with, result in any breach of, constitute a default (or an event which with the giving of notice or lapse of time, or both, would become a default) under,

11

the execution, delivery and performance of this Agreement by the Subscriber have been duly authorized by all necessary corporate action on the part of the Subscriber. This Agreement, when executed and delivered by the Subscriber, assuming due authorization, execution and delivery by the other parties thereto, constitutes the legal, valid, binding and enforceable obligations of the Subscriber in accordance with the terms hereunder, except as enforcement may be limited by general principles of equity, whether applied in a court of Law or a court of equity, and by applicable bankruptcy, insolvency and similar Law affecting creditors' rights and remedies generally.

## 6. COVENANTS

- Except as otherwise expressly provided in this Agreement or with the prior written consent of the Subscriber(which shall be provided promptly and not be unreasonably withheld), from the Effective Date until the earlier of the Closing Date and the valid termination of this Agreement in accordance with Section 7 (the "Interim Period"), the Company and Ningbo Geely shall, and each of the Company and Ningbo Geely shall cause all Group Companies to, to the extent permitted by applicable Laws:
  - 6.1.1 conduct their business in the ordinary course of business consistent with past practice;
  - preserve (i) the present business operations, organization (including officers and key employees) and goodwill of the Group Companies in the ordinary course of business consistent with past practice, and (ii) the present relationships with Persons having business dealings with the Group Companies (including customers and suppliers) in the ordinary course of business and consistent with past practice;
  - maintain (i) all of the assets and properties of, or used by, the Group Companies in their current condition consistent with past practice, except for ordinary course lease expiration, and ordinary wear and tear, and (ii) insurance upon all of the properties and assets of the Group Companies in such amounts and of such kinds comparable to that in effect on the date of this Agreement;
  - (i) maintain the books, accounts and records of the Group Companies in the ordinary course of business consistent with past practice, (ii) continue to collect accounts receivable and pay accounts payable using normal procedures and without discounting or accelerating payment of such accounts in the ordinary course of business consistent with past practice, and (iii) comply with all contractual and other obligations of the Group Companies in all material respects; and
  - 6.1.5 comply in all material respects with all applicable Laws.

12

- Without limiting the generality of Section 6.1, except as otherwise expressly provided in this Agreement or with the prior written consent of the Subscriber, during the Interim Period, the Company shall not to, and the Company and Ningbo Geely shall cause all Group Companies not to, to the extent permitted by applicable Laws:
  - declare, set aside, make or pay any dividend or other distribution in respect of the share capital of, or other equity interests or ownership interests in, any Group Company or repurchase, redeem or otherwise acquire any outstanding share capital, equity interests or other securities of, or other ownership interests in, any Group Company;

- transfer, issue, sell, pledge, encumber or dispose of any equity interests, share capital or other securities of, or other ownership interests in, any Group Company or grant options, warrants, calls or other rights to purchase or otherwise acquire equity interests, share capital or other securities of, or other ownership interests in, any Group Company;
- effect any recapitalization, reclassification, share split, share combination or like change in the capitalization of any Group Company, or amend the terms of any outstanding securities of any Group Company;
- amend the articles of association or equivalent organizational or governing documents of any Group Company, excluding (i) what is necessary to consummate the transactions contemplated by this Agreement, and (ii) the necessary updates to accommodate the requirement of PRC Company Law and other applicable Laws, provided that the amendments shall be made to the minimum extent required by Law and the Governmental Authorities and shall not substantially prevent or delay the consummation of the transactions contemplated by this Agreement;
  - except for those conducted in the ordinary course of business consistent with past practice, (i) increase the salary or other compensation of any director, officer or employee of any Group Company, (ii) grant any bonus, benefit or other direct or indirect compensation to any director, officer, employee or consultant, (iii) increase the coverage or benefits available under any (or create any new) severance pay, termination pay, vacation pay, company awards, salary continuation for disability, sick leave, deferred compensation, bonus or other incentive compensation, insurance, pension or other employee benefit plan or arrangement made to, for, or with any of the directors, officers, employees, agents or representatives of any Group Company or otherwise modify or amend or terminate any such plan or arrangement or (iv) enter into or amend any employment, deferred compensation, severance, special pay, consulting,

non-competition or similar agreement or arrangement with any directors, officers or employees of any Group Company;

subject to any lien or otherwise encumber or permit, allow or suffer to be encumbered, any of the properties or assets (whether tangible or intangible, including Intellectual Properties) of, or used by, any Group Company, other than Permitted Encumbrance or other than in the ordinary course of business consistent with past practice;

13

6.2.5

- acquire any properties or assets by any Group Company with an amount in excess of RMB 500,000 or sell, assign, 6.2.7 license, transfer, convey, lease or otherwise dispose of any of properties or assets of, or used by, any Group Company with an amount in excess of RMB 500,000, other than in the ordinary course of business consistent with past practice;
- enter into or agree to enter into any merger or consolidation by any Group Company with any corporation or other entity, or engage any Group Company in any new business;
- 6.2.9 invest in, make a loan, advance or capital contribution to, or otherwise acquire the securities, of any other Person by any Group Company with an amount in excess of RMB 500,000;
- 6.2.10 cancel or compromise any debt or claim or waive or release any material right of any Group Company with an amount in excess of RMB 500,000;
- incur, assume, alter, amend or modify any indebtedness for borrowed money, or guarantee thereof, or issue any debt securities with an amount in excess of RMB 500,000, except in the ordinary course of business consistent with past practice;
- 6.2.12 enter into any commitment for capital expenditures of any Group Company in excess of RMB 500,000 for any individual commitment;
- 6.2.13 make a change in the accounting or tax reporting principles, methods or policies of any Group Company;
- enter into any Contract (i) that restrains, restricts, limits or impedes the ability of any Group Company to compete with or conduct any business or line of business in any geographic area or solicit the employment of any persons; (ii) under which the transaction amount exceeds RMB 500,000, except in the ordinary course of business consistent with past practice;

- enter into any Contract that would be a Material Contract if such Contract had been entered into prior to the date of 6.2.15 this Agreement or materially amend, modify, renew (other than any automatic renewal in accordance with the relevant contractual terms), consent to the termination of, or waive any material rights under, any Material Contract;
- commence any Action for a claim of more than RMB 500,000 or settle or compromise any pending or threatened Action for an amount in excess of RMB 500,000 or that would impose any material restrictions on the business or operations of any Group Company;
- 6.2.17 effect or commence any liquidation, dissolution, winding-up, scheme of arrangement, restructuring, reorganization or similar transaction involving any Group Company;

14

- permit any Owned Intellectual Property of any Group Company to lapse or to be abandoned, dedicated, or disclaimed, fail to perform or make any applicable filings, recordings or other similar actions or filings, fail to pay all required fees and taxes required or advisable to maintain and protect its interest in such Owned Intellectual Property, or grant or license or transfer to any third party any such Owned Intellectual Property, except grants of non-exclusive licenses of Intellectual Property or in the ordinary course of business consistent with past practice or except where such permission, failure or grant would not result in a Material Adverse Effect;
- 6.2.19 take any action that would adversely affect the ability of the Parties to consummate the transactions contemplated by this Agreement; or
- agree to do anything (i) prohibited by this Section 6.2, (ii) that would make any of the representations and warranties of the Company or Ningbo Geely in this Agreement untrue or incorrect in any material respect or could result in any of the conditions to the Closing not being satisfied or (iii) that would be reasonably expected to have a Material Adverse Effect.

#### 6.3 Access to Information

In each case subject to restrictions under applicable Laws and the Group Companies' confidentiality obligation toward third parties,

during the Interim Period, the Company shall, and Ningbo Geely shall, by taking all necessary measure in its capacity as a shareholder of the Company, assist the Company to, provide the Subscriber with (i) unaudited monthly financial statements and (ii) operating or management information of the Group Companies reasonably required by the Subscriber. In addition, the Company shall and Ningbo Geely shall, by taking all necessary measure in its capacity as a shareholder of the Company, assist the Company to promptly provide the Subscriber with other information reasonably requested by the Subscriber, including any information requested in connection with the Subscriber's financing for the Equity Transfer.

as soon as reasonably practicable, but in no event later than twenty (20) days after the end of each calendar month

the Company and its Subsidiaries shall allow the Subscriber and its accountants, counsel, financial advisors and other representatives, at their own cost, reasonable access, during normal business hours upon prior notice during the Interim Period, to the Group Companies' respective properties and facilities (including all real property and the buildings, structures, fixtures, appurtenances and improvements erected, attached or located thereon), books, financial information, contracts and records of the Group Companies and, during such period, shall as soon as reasonably possible furnish such information concerning the businesses, properties and personnel of the Group Companies as the Subscriber shall reasonably request; provided, however, that such access in each case shall not disrupt the Group Companies' operations.

#### 6.4 Notification of Certain Matters

Each Party shall give notice to the other Parties as promptly as reasonably practicable upon becoming aware of (a) any fact, change, condition, circumstance, event, occurrence or non-occurrence that has caused or is reasonably likely to cause any representation or warranty in this Agreement made by it to be untrue or inaccurate in any respect at any time after the date hereof and prior to the Closing, (b) any failure on its part to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it hereunder or (c) the institution of or the threat of institution of any Action against it that may adversely affect the transactions contemplated hereby or related to this Agreement or the transactions contemplated hereby; (d) any notice or other communication from any Governmental Authority in connection with the transactions contemplated by this Agreement, or (e) any notice or other communication from any Person alleging that the consent of such Person is or may be required in connection with the transactions contemplated by this Agreement; provided that the delivery of any notice pursuant to this Section 6.4 shall not limit or otherwise affect the remedies available hereunder to the party receiving such notice, or the representations or warranties of, or the conditions to the obligations of, the parties hereto.

# 6.5 Further Assurances

Each party shall use its reasonable best efforts to take, or cause to be taken, all actions necessary or appropriate to (i) consummate the transactions contemplated by this Agreement and (ii) cause the fulfillment at the earliest practicable date of all of the conditions to their respective obligations to consummate the transactions contemplated by this Agreement. Without limiting the generality of the foregoing, prior to and at the Closing Date, each Party shall cooperate with the other Parties without any further consideration to make all filings with, and to obtain all consents of, any Governmental Authority or any other Person under any permit, license, agreement, indenture or other instrument (including any consents), and to take all such other actions as such Party may reasonably be requested to take by the other party from time to time, consistent with the terms of this Agreement, in order to effectuate the provisions and purposes of this Agreement and the transactions contemplated hereby.

# 7. <u>TERMINATION</u>

- 7.1 This Agreement may be terminated at any time prior to the Closing:
  - (a) by the mutual written consent of the Parties;
  - (b) by any Party, by written notice to the other Parties, if the ETA is terminated;
  - by any Party, by written notice to the other Parties, if the Closing has not occurred on or before the Long Stop Date, provided that the right to terminate this Agreement pursuant to this Section 7.1(c) shall not be available to the Party seeking to terminate if such Party is in breach of, or has breached, any of its obligations under this Agreement, which breach has been a material cause of the failure of the Closing to be consummated on or before such date;

16

- (d) by any Party, by written notice to the other Parties, if the consummation of the transactions contemplated by this Agreement would violate any non-appealable final Governmental Order;
  - by written notice from the Subscriber if there has been a breach of any representation, warranty, covenant or agreement on the part of the Company or Ningbo Geely contained in this Agreement such that the conditions set forth in Section 4.1 would not be satisfied and such breach or failure of condition is not curable or, if curable, is not cured prior to the earlier
- (e) of (i) 30 days after written notice thereof is given by the Subscriber to the Company or Ningbo Geely or (ii) the Long Stop Date; provided that the Subscriber shall not have the right to terminate this Agreement pursuant to this Section 7.1(e) if the Subscriber is then in breach of any of its representation, warranty, covenant or agreement contained in this Agreement that would cause the conditions set forth in Section 4.2 not to be satisfied; or
- by written notice from the Company or Ningbo Geely if there has been a breach of any representation, warranty, covenant or (f) agreement on the part of the Subscriber contained in this Agreement such that the conditions set forth in Section 4.2 would not be satisfied and such breach or failure of condition is not curable or, if curable, is not cured prior to the earlier of (i) 30

days after written notice thereof is given by the Company or Ningbo Geely, as applicable, to the Subscriber or (ii) the Long Stop Date; provided that the Company or Ningbo Geely, as applicable, shall not have the right to terminate this Agreement pursuant to this Section 7.1(f) if the Company or Ningbo Geely, as applicable, is then in breach of any of its representation, warranty, covenant or agreement contained in this Agreement that would cause the conditions set forth in Section 4.1 not to be satisfied.

In the event of termination by any Party pursuant to Section 7.1 above, written notice thereof shall promptly be given to the other Parties and the transaction contemplated hereunder shall be terminated without further action by any Party; provided, however, that (a) the provisions of this Section 7 (*Termination*), Section 9 (*Confidentiality*), Section 10 (*Notice*), Section 11 (*Governing Law and Dispute Resolution*), Section 12 (*Taxes, Costs and Expenses*) and Section 13 (*Miscellaneous*) shall survive such termination,(b) the Parties shall take all necessary actions (including without limitation to cancelling or modifying any government registration, filings or other documents that have been conducted before such termination) to restore all Parties to their original position and status they were in immediately prior to the execution of this Agreement and the Party which is in breach of the Agreement (if any) shall be responsible for the relevant fees and Losses, and (c) nothing contained herein shall relieve any liabilities of any Party for breach of this Agreement that occurred prior to the termination hereof.

# 8. INDEMNIFICATION

7.2

8.1

9.1

From and after the Closing, each of Ningbo Geely and the Company (each, an "Indemnifying Party") hereby agrees to, jointly and severally, indemnify and hold the Subscriber, its Affiliates and their respective directors, officers, employees, agents, successors and assigns (collectively, the "Indemnified Parties" and each, an "Indemnified Party") harmless from and against any and all Losses which arise out of or result from: (i) any material breach, violation or nonperformance of any covenant or agreement of such Indemnifying Party under this Agreement, and/or (ii) any material breach of any of the representations or warranties made by such Indemnifying Party under this Agreement. For the avoidance of doubt, nothing in this Section released the Indemnified Parties to prove their Losses.

17

Notwithstanding any other provisions of this Agreement to the contrary, for any indemnification claim made by an Indemnified Party against any Indemnifying Party under this Section 8, the Indemnifying Party shall not have the right of recourse against any of the Group Companies in any event.

# 9. <u>CONFIDENTIALITY</u>

Subject to Sections 9.2 and 9.3, each Party undertakes with the other Parties that it shall treat as strictly confidential all information received or obtained by it, its Affiliates or the directors, officers, employees, agents, advisers or other representatives of such Party or its Affiliates (collectively, such Party's "Representatives") in connection with entering into or performing this Agreement, including information relating to the existence and provisions of this Agreement, the negotiations leading up to this Agreement, the materials and information acquired by the Representatives through their due diligence of the Group Company, the subject matter of this Agreement or the business or affairs of the other Parties or any Subsidiary of the other Parties (collectively, "Confidential Information"), and that it shall not at any time hereafter disclose or divulge to any Person any Confidential Information and shall use its best endeavor to prevent the publication or disclosure of any Confidential Information. Each Party also undertakes with the other Parties that neither it nor any of its Representatives shall trade or engage in any derivative or other transaction on the basis of such Confidential Information in violation of any applicable Laws.

Confidential Information shall not include any information that is (i) previously known on a non-confidential basis by the receiving Party or its Representatives, (ii) in the public domain through no fault of such receiving Party or its Representatives, (iii) received from a Person other than any of the other Parties or their respective Representatives, so long as such Person was not, to the best knowledge of the receiving Party, subject to a duty of confidentiality to such other Party, or (iv) developed independently by the receiving Party without reference to Confidential Information of the disclosing Party.

- 9.3 Notwithstanding Section 9.1:
  - any Party may disclose Confidential Information to the extent that such disclosure (i) is required under applicable Laws or any judicial or regulatory process or (ii) is requested by any Governmental Authority or other regulatory body,

including the rules and requirements of any securities exchange, or (iii) is made in relation to the arbitration as set forth in Section 11; provided, that, in the case of the foregoing (i) and (ii), such Party shall, to the extent permitted by Law and so far as it is practicable, provide the other Parties with prompt notice of such requirement or request and cooperate with the other Parties at such other Parties' request and cost to enable such other Parties to seek an appropriate protection order or remedy; and

18

any Party may disclose Confidential Information to its Representatives on a need-to-know basis; provided, that such Party shall use reasonable best efforts to ensure that each such Person to which it discloses Confidential Information strictly abides by the confidentiality obligations hereunder and shall be responsible for any breach of confidentiality obligations by such Person.

Except as may be required by applicable Law, the press release announcing the execution of this Agreement shall be issued only in such form as shall be agreed among the Parties. Thereafter, at any time prior to the earlier of the Closing Date and the valid termination of this Agreement pursuant to Section 7, except as may be required by applicable Law, each Party shall consult with the other Parties before it or its Affiliates issues any press release, has any communication with the press, making any other public statement with respect to this Agreement and the transactions contemplated hereunder, and shall provide each other a reasonable opportunity to review and comment on (and consider such proposed comments in good faith), such press releases, communication or public statement.

# 10. NOTICE

9.4

10.1

Any notice given by a Party to the other Parties under this Agreement shall be deemed validly served by hand delivery or by prepaid registered mail sent through the post (airmail if to an overseas address) or by email transmission to its address given herein or such other address as may from time to time be notified for this purpose and any notice served by prepaid registered letter shall be deemed to have been served 48 hours (72 hours in the case of a letter sent by airmail to an address in another country) after the time at which it was posted, any notice served by email transmission shall be deemed to have been served at the time of the email transmission, it shall be sufficient (in the case of service by hand and prepaid registered letter) to prove that the notice was properly addressed and delivered or posted, as the case may be, and in the case of service by email transmission to prove that there is no system-generated non-delivery notice received by the sender.

Each notice, demand or other communication given or made under this Agreement shall be in writing and delivered or sent to the relevant Party at its address or telex number or email address set out hereinafter (or such other address or email address as the addressee has by five (5) days' prior written notice specified to the other Parties):

To Zhejiang ZEEKR:		
To Ningbo Geely:		
To range deciy.		
To Company:		

# 11. GOVERNING LAW AND DISPUTE RESOLUTION

19

The validity, construction, implementation and interpretation of this Agreement shall be governed by and construed in accordance with the Laws of the PRC.

# 11.2 Dispute resolution

#### 11.2.1 Consultations

In the event of a dispute arising out of or in connection with the interpretation or implementation of this Agreement (the "**Dispute**"), the Parties shall attempt in the first instance to resolve the Dispute in good faith by friendly consultations between the Parties. If the Dispute cannot be resolved by any such method within thirty (30) Business Days after the commencement of consultation, then any Party may submit the Dispute to arbitration.

#### 11.2.2 Arbitration

Any Dispute referred to arbitration shall be finally resolved by arbitration administered by the China International Economic and Trade Arbitration Commission Shanghai Sub-Commission("CIETAC"), which shall be held in Shanghai and conducted

- (a) in accordance with the CIETAC arbitration rules in force when the notice of arbitration is submitted by a Party, which rules are deemed to be incorporated by reference into this section, by three (3) arbitrators appointed in accordance with the said rules, and in such case:
  - (i) all proceedings in any such arbitration shall be conducted in English;
  - (ii) all of the arbitrators shall be fluent in English; and
  - (iii) only the English text of this Agreement shall be referred to by the arbitrators in the course of the arbitration proceedings.
- (b) Should a Dispute be submitted to arbitration, all Parties shall in all other respects, except in the event of termination, continue to perform their respective obligations in accordance with this Agreement.

The arbitration award shall be final and binding on all Parties. The costs of arbitration shall be borne by the losing Party or as otherwise determined by the arbitration tribunal. Any award of the arbitrator tribunal shall be enforceable by any (c) court having jurisdiction over a Party against which the award has been rendered, or in any place where assets of a Party

against which the award has been rendered are located, and will be enforceable in accordance with applicable Laws in the jurisdiction in which such enforcement is sought, and the terms of applicable treaties and international conventions.

20

#### 12. TAXES, COSTS AND EXPENSES

Unless as otherwise expressly specified by this Agreement, each Party shall bear its own taxes, costs and expenses incurred in connection with negotiation, preparation, execution and performance of this Agreement, including but not limited to, stamp duty, income tax, financial and legal due diligence costs, valuation fees or other legal fees.

# 13. MISCELLANEOUS

- 13.1 Neither Party shall be entitled to a duplicative compensation for the same Loss under this Agreement and the ETA.
- This Agreement shall take effect upon signing by the Parties on the date first written above and shall be binding upon and ensure for the benefit of the estates, successors and/or permitted assigns of the Parties.
- Each of the Parties hereto undertakes to the other Parties that it shall do all such actions and things and execute all such deeds and documents as may be necessary or desirable to carry into effect or to give legal effect to the provisions of this Agreement and the transactions hereby contemplated.

This Agreement constitutes the whole agreement between the Parties and supersede and terminate any previous agreements or 13.4 arrangements between them relating to the subject matter hereunder; it is expressly declared that no variations hereof shall be effective unless made in writing and signed by duly authorized representatives of the Parties. If any provision or part of a provision of this Agreement shall be, or be found by any authority or court of competent jurisdiction 13.5 to be, invalid or unenforceable, such invalidity or unenforceability shall not affect the other provisions or parts of such provisions of this Agreement, all of which shall remain in full force and effect. The Parties to this Agreement are independent of each other. Each Party shall exercise its rights and assume its obligations 13.6 hereunder independently. Any action or judgment by either Party shall neither affect the other Parties nor affect the rights and obligations of the other Parties hereunder or the force of any section hereof concerning the other Parties. No Party may assign this Agreement and/or any of its rights, interests, or obligations hereunder without the prior written 13.7 approval of the other Parties. This Agreement may be executed in one or more counterparts, and by the Parties on separate counterparts, but shall not be effective until each Party has executed at least one counterpart and each such counterpart shall constitute an original of this 13.8 Agreement but all the counterparts shall together constitute one and the same instrument. This Agreement is made in English. If a Chinese translation of this Agreement is delivered it is for information purposes only 13.9 and does not constitute a definitive and legally binding agreement. For the avoidance of doubt, in case there is any conflict between the Chinese translation and the English version, the English version shall prevail. 21 < Remainder of this page intentionally left blank> 22 Signature page to the SUBSCRIPTION AGREEMENT IN WITNESS WHEREOF the Parties have hereunto caused this Agreement to be duly executed as of the day and year first above written. On behalf of ZHEJIANG ZEEKR INTELLIGENT TECHNOLOGY CO., LTD. (浙江极氪智能科技有限公司) (Company seal) Signature:/s/ Conghui An Name: Conghui An Title: **CEO** 

IN WIT written.	NESS WHEREOF the Parties have hereunto caused this Agreement to be duly executed as of the day and year first above
On behal	lf of
NINGBO	O GEELY AUTOMOBILE INDUSTRY CO., LTD. (宁波吉利汽车实业有限公司)
(Compa	ny seal)
Signature	e:/s/ Gan Jiayue
Name:	Gan Jiayue
Title:	Legal Representative / Director
	Signature page to the SUBSCRIPTION AGREEMENT
	Signature page to the SUBSCRIF HOW AGREEMENT
	NESS WHEREOF the Parties have hereunto caused this Agreement to be duly executed as of the day and year first above
written.	
On behal	If of
On ochai	
LYNK &	& CO AUTOMOTIVE TECHNOLOGY CO., LTD. (领克汽车科技有限公司)
(Compa	ny seal)
Signature	e:/s/ Gan Jiayue
Name:	Gan Jiayue
Title:	Legal Representative / Director